Institute for Monetary and Financial Stability

BALINT TATAR

Has the Reaction Function of the European Central Bank Changed Over Time?

Institute for Monetary and Financial Stability GOETHE UNIVERSITY FRANKFURT

WORKING PAPER SERIES NO. 183 (2023)

This Working Paper is issued under the auspices of the Institute for Monetary and Financial Stability (IMFS). Any opinions expressed here are those of the author(s) and not those of the IMFS. Research disseminated by the IMFS may include views on policy, but the IMFS itself takes no institutional policy positions.

The IMFS aims at raising public awareness of the importance of monetary and financial stability. Its main objective is the implementation of the "Project Monetary and Financial Stability" that is supported by the Foundation of Monetary and Financial Stability. The foundation was established on January 1, 2002 by federal law. Its endowment funds come from the sale of 1 DM gold coins in 2001 that were issued at the occasion of the euro cash introduction in memory of the D-Mark.

The IMFS Working Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Institute for Monetary and Financial Stability Goethe University Frankfurt House of Finance Theodor-W.-Adorno-Platz 3 D-60629 Frankfurt am Main www.imfs-frankfurt.de | info@imfs-frankfurt.de

Has the Reaction Function of the European Central Bank Changed Over Time?

Balint Tatar

Goethe University Frankfurt German Council of Economic Experts

First version: December 30, 2022 This version: May 2, 2023

Abstract

I have assessed changes in the monetary policy stance in the euro area since its inception by applying a Bayesian time-varying parameter framework in conjunction with the Hamiltonian Monte Carlo algorithm. I find that the estimated policy response has varied considerably over time. Most of the results suggest that the response weakened after the onset of the financial crisis and while quantitative measures were still in place, although there are also indications that the weakening of the response to the expected inflation gap may have been less pronounced. I also find that the policy response has become more forceful over the course of the recent sharp rise in inflation. Furthermore, it is essential to model the stochastic volatility relating to deviations from the policy rule as it materially influences the results.

Keywords: Monetary policy rules, Bayesian time-varying parameter estimation, unconventional monetary policy, Hamiltonian Monte Carlo *JEL classification:* E52, C11, C22, C51

^{*}E-Mail: balint.tatar@googlemail.com. The main idea underlying this paper is partially based on the assessments which I carried out as a member of the scientific staff at the German Council of Economic Experts (GCEE) in cooperation with Leonard Salzmann and some results were later made public in the Annual Report 2021/22 of the GCEE. I am grateful to Michael Binder, Philipp Harms, Thomas Otter, Leonard Salzmann and Volker Wieland for valuable comments. The views expressed in this paper are solely my own and do not necessarily reflect the views of the GCEE.

1. Introduction

Simple policy rules have often been applied to assess the stance of monetary policy since the seminal contribution made by Taylor (1993). Several studies exist – mainly for the US – which estimate simple rules using regime-switching models or time-varying parameter frameworks to find evidence of changes in monetary policy over time (e.g. Sims (2001), Cogley and Sargent (2001, 2005), Boivin (2006), Kim and Nelson (2006), Sims and Zha (2006)).¹ However, the latter strand of the literature appears to be limited for the euro area.

In this paper I expand the available literature by estimating the evolution of the monetary policy response in the euro area since the inception of monetary union. I apply a Bayesian non-linear framework in combination with the Hamiltonian Monte Carlo (HMC) algorithm, which is superior to other existing sampling methods owing to its efficiency and unique diagnostic features. This modelling framework allows for both time-varying parameters and heteroscedasticity within a euro-area-specific monetary policy rule and does not require any linear approximation or the application of the Kalman filter. I scrutinise to what extent the non-standard monetary policy measures have influenced the policy response in the aftermath of the financial crisis and attempt to shed light on the role that monetary policy may have played in the run-up to the recent sharp rise in inflation in the euro area. Furthermore, I evaluate the European Central Bank's (ECB) recent policy response to the surge in inflation.

A remarkable amount of earlier literature provides point estimates mostly of Taylor-type or ad-hoc reaction functions of the ECB.² Gorter et al. (2010) applied a rolling-window OLS approach to assess possible changes in the ECB's reaction function during the financial crisis, while Gerlach and Lewis (2014a, 2014b) used a smooth transition framework to allow for shifts between two different parameter sets for the estimated reaction function. Rostagno et al. (2019) estimated policy rules for

 $^{^{1}\}mathrm{Earlier}$ studies such as e.g. Judd et al. (1998), Clarida et al. (2000) or Orphanides (2004) used subsample analysis for this purpose.

 $^{^{2}\}mathrm{See},$ for example, the studies cited in Blattner and Margaritov (2010) or Gerlach and Lewis (2014b).

the period from 1999 to 2008 and found a more forceful response when inflation was above its 2 percent target than when it was below this level. Maih et al. (2021) estimate, among other things, a regime-switching version of the Smets-Wouters DSGE model up to the end of 2014.2. They allow for an endogenous switch of the coefficient on inflation within a smoothed Taylor-type policy rule depending on whether inflation is above or below its target. They conclude that until mid-2014 the ECB responded more forcefully to inflation of above 1.9 percent than it did to inflation below that level. Furthermore, based on their estimation results up to mid-2014, they show by likelihood comparison that the ECB conducted a symmetric policy during the period after mid-2014. One drawback with this approach is that it does not enable the policy response to be directly estimated while quantitative easing was in place. In addition, only the switch in the ECB's response to inflation is taken into account, although its response to other economic aggregates within the policy rule may also have changed, which would probably alter the relevant results. Furthermore, this methodology does not allow continuously changing responses during the assessed period to be estimated. Paloviita et al. (2021) use GMM and a rollingwindow approach to estimate a large variety of Taylor-type policy rules for the period from 1999 to 2014. They conclude that policy responses have been asymmetric, with the responses to inflation overshooting its target being more forceful than those to inflation undershooting.

This paper departs from the framework applied in Orphanides and Wieland (2013) (OW henceforth) and Bletzinger and Wieland (2017) (BW henceforth). They characterise the ECB's reaction function by assuming that the policy rate is set according to the following rule:

$$i_t = i_{t-1} + \theta_1(\pi_{t+3|t} - \pi^*) + \theta_2(q_{t+2|t} - q_{t+2|t}^*)$$
(1)

Here, *i* denotes the policy rate, π the inflation rate, π^* the inflation target, *q* the growth rate of real GDP and q^* the growth rate of real potential GDP. The subscript t+i|t refers to the expected value at time *t* of a particular variable *i* quarters ahead.

A change in the policy rate compared with the level in the previous period occurs when expected inflation deviates from the central bank's inflation target or expected GDP growth deviates from expected potential growth. OW show that by setting the parameters θ_1 and θ_2 to 0.5 each and the inflation target to 1.5 or 2.0, the policy rate prescriptions under the policy rule using data from the Survey of Professional Forecasters (SPF) and the European Commission are very similar to the policy rate set by the ECB over time. This rule for the euro area was estimated by, among others, Smets (2010), BW and Hartmann and Smets (2018) (HS henceforth), who applied the following framework:³

$$\Delta i_t = c + \theta_1 \pi_{t+3|t} + \theta_2 (q_{t+2|t} - q_{t+2|t}^*) + \epsilon_t \tag{2}$$

where Δi_t is the change in the policy rate compared with the previous period, c is a constant and ϵ_t denotes the error term with $\epsilon_t \sim N(0, \sigma^2)$. Consequently, the estimated inflation target is given by $\pi^* = c/\theta_1$. BW estimated θ_1 and θ_2 at 0.49 and 0.40, respectively, and the inflation target at 1.72 in their baseline specification. HS estimated the above parameters with staff projections of the ECB/Eurosystem instead of SPF forecasts for the period from 1999.1 to 2018.1 in their baseline specification at 0.34, 0.37 and 1.81, respectively. To assess the stability of the parameter estimates over time HS carried out the estimation with a shortened sample until 2012.2. They report that the estimation results differ only slightly from the estimates based on the full sample suggesting that the regression coefficients should remain stable over time.

The latter studies have major shortcomings, though. Firstly, they partly disregard that the reaction function of a central bank can vary over time. Even if the regression is estimated on a shorter sample as in HS, the estimated inflation target will change as well, although the monetary policy strategy remained the same over the entire sample period. This will possibly bias the estimates of the reaction coefficients. Secondly, these studies do not take account of the effect that quantitative

 $^{^{3}}$ Smets (2010) regressed the interest rate on its lag but estimated fairly high values from 0.89 (baseline setup) up to 0.97, depending on the model specification.

easing has on the key policy rate. Although HS, for example, used a time series for their estimation which combined the changes in the MRO up to 2008.3 with the changes in the deposit rate from 2008.4 onwards, the latter series may not fully capture the magnitude of the non-standard measures adopted by the ECB in the aftermath of the financial crisis. Thirdly, none of these studies applied a time-varying parameter framework, which would provide accurate estimates of the central bank's response in the euro area over time and may therefore result in wrong inference if comparing different time periods.

In this paper I attempt to remedy these shortcomings by providing a consistent framework for measuring continuous changes in the monetary policy response up to the present. I find that the estimated policy response has varied considerably over time. In particular, most of the results suggest that the policy response weakened after the beginning of the financial crisis and while quantitative measures were in place. There are also indications, however, that the weakening of the response to the expected inflation gap may have been less pronounced, depending on the data set used for estimation purposes. I also find that the policy response has strengthened during the recent surge in inflation. Moreover, the introduction of stochastic volatility associated with deviations from the policy rule is also essential, as it materially affects the results.

The rest of this paper is organized as follows. Sections 2 describes the model setup and the estimation approach. Section 3 provides a description of the data. Section 4 discusses the baseline results. Section 5 presents robustness checks and extensions of the baseline framework. Section 6 concludes the paper.

2. A Bayesian time-varying parameter representation

To describe the law of motion of the ECB's response to macroeconomic aggregates I apply a Bayesian non-linear regression framework with time-varying parameters which is well suited to capture the gradual change in the response of monetary policy over time. I estimate the policy rule from OW, similarly to BW, but treat the reaction coefficients $\theta_{1,t}$ and $\theta_{2,t}$ as being random to obtain the posterior densities of both coefficients for all points in time t in the period from 1999.2 to 2021.2 for the euro area.

2.1. The model

The model can be represented by means of a non-linear state space where the observation equation is given by

$$y_t = f(X_t, \hat{\theta}_t) + \epsilon_t \qquad t = 1, ..., T.$$
(3)

 y_t is a scalar variable, $\tilde{\theta}_t$ is a $K \times 1$ dimensional vector of parameters to be estimated, X_t is a vector of predetermined or exogenous variables and ϵ_t is the error term. For this case, y_t is set to be the change in the key policy rate at time t, Δi_t , the left-hand side of the policy rule in OW. The time-varying coefficients are treated as the hidden state vector in the state space. To describe the dynamics of the reaction coefficients, θ_1 and θ_2 , over time, I postulate a driftless random walk process:

$$\theta_{i,t} = \theta_{i,t-1} + \nu_{i,t} \qquad i = 1,2 \tag{4}$$

where I assume that $\nu'_t := (\nu_{1,t}, \nu_{2,t})$ is i.i.d. with a covariance matrix Q. This allows to characterize the evolution of $\theta_t := (\theta_{1,t}, \theta_{2,t})$ over time as follows:

$$p(\theta_{t+1}|\theta_t, \Omega) \propto h(\theta_{t+1}|\theta_t, \Omega) \tag{5}$$

where $p(\theta_{t+1}|\theta_t, \Omega)$ is the conditional probability density of the state vector for the next period, θ_{t+1} , given the state vector in the current period, θ_t , as well as Ω , denoting the covariance matrix of the state space to be defined below and

$$h(\theta_{t+1}|\theta_t, \Omega) \sim N(\theta_t, Q). \tag{6}$$

Furthermore, I also assume that the error terms, $(\epsilon_t, \nu'_t)'$, are i.i.d. normal random variables with zero mean and a covariance matrix

$$\Omega := \begin{bmatrix} \epsilon_t \\ \nu_t \end{bmatrix} \begin{bmatrix} \epsilon_t & \nu'_t \end{bmatrix} = \begin{bmatrix} \sigma^2 & C' \\ C & Q \end{bmatrix}$$
(7)

where σ^2 is the variance of ϵ . Q is the covariance matrix of the innovations of the state equation and C denotes the cross covariances. To derive the joint likelihood function let us first define $\tilde{\theta}_t := (\theta_t, \pi^*)$, the vector consisting of the states at time t and the inflation target. In the following I will denote the partial histories of the states and the observed variable by $\theta^t := [\theta_1, ..., \theta_t]$ and $Y^t := [y_1, ..., y_t]$, respectively, for all t = 1, ..., T. Consequently, the partial history of $\tilde{\theta}$ is defined as $\tilde{\theta}^t := [\theta_1, ..., \theta_t, \pi^*]$. To derive the joint posterior density function one can exploit first the probabilistic structure of the random walk process characterizing the time-varying parameters:

$$p(\theta^T | \Omega) = p(\theta_0 | \Omega) \prod_{t=0}^{T-1} p(\theta_{t+1} | \theta_t, \Omega).$$
(8)

According to Bayes' law, the posterior density can be expressed as the product of the likelihood and the joint prior:

$$p(\tilde{\theta}^T, \Omega | Y^T) \propto p(Y^T | \tilde{\theta}^T, \Omega) p(\tilde{\theta}^T, \Omega).$$
(9)

Since the observations are conditionally independent and by assuming that the cross covariance is zero, the likelihood can be factored as follows:

$$p(\tilde{\theta}^T, \Omega | Y^T) \propto p(\tilde{\theta}^T, \Omega) \prod_{t=0}^T p(y_t | \tilde{\theta}^T, \Omega) = p(\tilde{\theta}^T, \Omega) \prod_{t=0}^T p(y_t | \tilde{\theta}^t, \Omega).$$
(10)

The equation holds because the policy rate setting rule does not depend on future values of the reaction coefficients. Furthermore, given the probabilistic structure of θ stated above and since the central bank changes the policy rate only in response to the expected inflation gap and the expected growth gap, the posterior can be

expressed as follows:

$$p(\tilde{\theta}^T, \Omega | Y^T) \propto p(\tilde{\theta}^T, \Omega) \prod_{t=0}^T p(y_t | \tilde{\theta}_t, \Omega).$$
 (11)

Following Koop (2003) the likelihood function of the non-linear univariate regression model can be written in vector form:

$$p(y|\tilde{\theta},\sigma) = \frac{1}{(2\pi\sigma^2)^{N/2}} \left\{ \exp\left[-\frac{1}{2\sigma^2} \{y - f(X,\tilde{\theta})\}' \{y - f(X,\tilde{\theta})\}\right] \right\}.$$
 (12)

Setting $f(X, \tilde{\theta}) = \theta_1(\pi^e - \pi^*) + \theta_2(q^e - q^{*,e})$ the likelihood will translate into

$$p(Y^{T}|\tilde{\theta}^{T},\Omega) = \prod_{t=0}^{T} \frac{1}{(2\pi\sigma^{2})^{1/2}} \left\{ \exp\left[-\frac{(y_{t} - f(X_{t},\tilde{\theta}_{t}))^{2}}{2\sigma^{2}}\right] \right\}$$
(13)

with
$$f(X_t, \tilde{\theta}_t) := \theta_{1,t}(\pi_{t+3|t} - \pi^*) + \theta_{2,t}(q_{t+2|t} - q_{t+2|t}^*).$$
 (14)

The joint prior can be further rewritten by factoring it into the conditional prior for θ^T and the joint density of the inflation target and the state space covariance:

$$p(\tilde{\theta}^T, \Omega) \equiv p(\theta^T, \pi^*, \Omega) = p(\theta^T | \pi^*, \Omega) p(\pi^*, \Omega).$$
(15)

I assume that the central bank response is characterized by a random walk process which does not depend on the inflation target. This may appear slightly restrictive, however it could be very challenging to specify a relation and consequently a conditional distribution for θ depending on π^* as well, if there exists any at all. Therefore, given the assumption that the central bank response is independent from π^* , the latter can be dropped from the conditional distribution of θ^T , so one can make directly use of equation (8). Furthermore, assuming that the initial state, θ_0 , and the covariance matrix of the state space are independent, the prior can be expressed as follows:

$$p(\theta^T | \pi^*, \Omega) p(\pi^*, \Omega) = p(\pi^*) p(\Omega) p(\theta_0) \prod_{t=0}^{T-1} p(\theta_{t+1} | \theta_t, \Omega).$$
(16)

To complete the model the prior distributions have to be specified. In the baseline setup, I assume a Gaussian distribution for the initial states:

$$p(\theta_0) \propto N(\bar{\theta}, \bar{P}) \tag{17}$$

with $\bar{\theta} = (0.5, 0.5)$ and a diagonal \bar{P} matrix. I set a relatively wide prior with entries on the main diagonal of \bar{P} being equal to $\bar{p}_{11} = \bar{p}_{22} = 4$ to let the data determine the initial value. The prior distribution of the inflation target is assumed to be $\pi^* \sim N(1.9, 0.15^2)$ which reflects the monetary policy strategy of the ECB, in particular that inflation is targeted to be "below, but close to 2 percent". For the prior distribution of the state space covariance matrix, Ω , I will assume that the cross covariance, C, is zero. Furthermore, the covariance matrix structure of the state equation innovations, Q, is assumed to be diagonal. Consequently, for the standard deviation of both random walk innovations, σ_{ν_1} and σ_{ν_2} , and for the standard deviation of the error term in the policy rule, σ , I will posit an inverse gamma distribution with an expected value of 0.2 and a relatively wide standard deviation of 0.2, respectively.⁴ I will use the above priors to first estimate the model on the training sample between 1999.2 and 2001.4. Subsequently, the results obtained using the latter sample will be set as priors for the estimation of the complete sample.

2.2. Simulation of the posterior density

The estimation of the policy rule from (1) in its original form in conjunction with time-varying coefficients poses a serious challenge as the observation equation is non-linear. A straightforward approach to address non-linearity issues would be to follow the setup in BW, yet to estimate equation (2) with a time-varying intercept and to rely on established estimation methods. However, this would result in identification problems and disregard at the same time that the estimated intercept depends on θ_1 . Furthermore, the implied inflation target would be time-varying as well. Assuming that θ_1 and θ_2 are time-varying requires the estimation of more than

⁴This corresponds to $\sim IG(\alpha, \beta)$ with $\alpha = 3$ and $\beta = 0.4$.

2*T* parameters. A readily available Bayesian estimation method capable of sampling from high-dimensional distributions, given that a posterior distribution can be derived, is the HMC algorithm. The latter approach has been widely applied in other fields of science owing to its efficiency and superior diagnostic features, see e.g. Neal (2011) and Betancourt (2017). The main idea of the HMC algorithm is to extend the parameter space by an auxiliary vector of momentum variables, α , to obtain the joint posterior density of $\tilde{\theta}^T$, Ω and α :

$$p(\tilde{\theta}^T, \Omega, \alpha | Y^T) \propto p(Y^T | \tilde{\theta}^T, \Omega, \alpha) p(\tilde{\theta}^T, \Omega, \alpha).$$
(18)

To each parameter to be estimated in the original model, one momentum variable α_i is assigned which is a priori independent of $\tilde{\theta}^T$, Ω and Y^T so that $p(\tilde{\theta}^T, \Omega, \alpha | Y^T) \propto$ $p(Y^T | \tilde{\theta}^T, \Omega) p(\tilde{\theta}^T, \Omega) p(\alpha)$. In the extended parameter space one can apply Hamiltonian dynamics, a well known concept from physics. The Hamiltonian equation describes the total energy of a frictionless mechanical system, in particular the position and the momentum of a moving particle. In this extended framework, the model parameters to be estimated are considered as the position of a particle. In a frictionless mechanical system the Hamiltonian equation in this extended parameter space allows to find an update proposal which is distant from the original position of the particle, that is, of the parameters to be estimated, while the acceptance rate remains very high, even if the dimension of the parameter space is high. For further details I refer to Neal (2011) and Betancourt (2017) who provide an excellent description of the methodology along with its superior features compared to standard random walk based sampling methods.

3. Data description

I use a quarterly data set for the estimation which is partially compiled from monthly data. As regards the change in the key policy rate I use the first difference of the main refinancing operations (MRO) rate of the ECB until 2008.2. From 2008.3 onwards, when the global financial crisis started to affect the euro area, I replace the latter series with the first difference of the shadow interest rate series. Thereby I account for the impact of non-conventional measures in periods when the effective lower bound was possibly binding. I also remedy a further shortcoming in the existing literature which does not take at all, or at least not properly, account of the effects of the quantitative measures carried out by the ECB. Although HS uses the deposit rate instead of the key policy rate from 2008.4 onwards, the latter series reflects the changes in the policy rate only partly, at best, as can be seen in Figure 1. Two different monthly shadow rate series are regularly estimated and updated

Figure 1: Interest rates, expected inflation and growth gap in the euro area



Notes: Plot (a): the dashed line shows the MRO rate and the dotted line the deposit facility rate of the ECB. The blue and the red lines represent the shadow interest rates based on the Krippner and the Wu-Xia datasets, respectively. Plot (b): the blue line shows the expected HICP inflation in t+3, while the red line represents the difference between the expected GDP growth and forecasted potential GDP growth in t+2.

by Krippner and Wu, respectively, based on their previous studies (Krippner (2013, 2015); Wu and Xia, (2017, 2020)). The series differ remarkably, therefore I will carry out all estimations with both series separately. In each case, also for the MRO, I use rates in the second month of the quarter to build the first differences, as the Survey of Professional Forecasters (SPF), where the expected HICP inflation and the expected GDP growth series stem from, are published at the end of the first month each quarter. Therefore, both explanatory variables could be considered as weakly exogenous. As in BW and HS, I use the 4-quarters-ahead forecasts from the most recent data point available for both explanatory variables. I use forecast data from the SPF, in particular, expected yearly HICP inflation in t+3 quarters and expected

yearly GDP growth in t + 2 quarters ahead owing to the different availability of the HICP and the GDP growth data series. To calculate the forecasted yearly potential growth rate on a quarterly basis I rely on the yearly estimates and projections of the European Commission as no publicly available estimates are provided by the ECB and transform the periodicity of the series by quadratic interpolation.⁵

Figure 1 shows that in normal times, when the central bank was not constrained by the zero lower bound and large scale asset purchases were not used as a standard policy instrument, the shadow rates differ only marginally from the MRO rate. However, with the onset of the financial crisis and the subsequent introduction of the non-standard measures the shadow rates fell significantly below the MRO rate. Although the gap between the shadow rates and the MRO rate closed in the first half of 2011, the subsequent measures taken to support both the private and the public sector in the euro area caused the shadow rates to diverge again significantly from the MRO rate. The gap widened even further with the introduction of the different asset purchase programmes until the end of 2016 when the pace of the asset purchases decreased. At this point, according to both data sets the shadow rate hovered around in negative territories well below the MRO rate and both shadow rate series showed a similar pattern. From the end of 2016 they exhibited a markedly different dynamics though. As a consequence of the subsequent reduction in net asset purchases the Krippner series has already signaled a tightening while according to the Wu-Xia series the shadow rate became even more negative. The different evolution continued until the termination of the asset purchase programmes at the end of 2018. From 2019.2 both series pointed again to an easing in accordance with the relaunch of the longer term refinancing operations and the restart of the asset purchases when inflation expectations started deteriorating again. Although at the beginning of the pandemic a further asset purchase programme was announced both shadow rate series fail to signal a monetary easing. This feature is possibly attributable to a substantial increase in the demand for liquidity due to the complete shutdown of

⁵I rely on the standard procedure provided in the software package Eviews.

the economy. In the first half of 2021 the shadow rates signaled again a tightening due to the repeated pandemic related shutdown of numerous economies in the euro area. Both the MRO and the deposit rate rate barely changed from 2015 onwards and clearly failed to reflect the impact of quantitative easing adequately. Therefore, the application of shadow rates for the estimation is warranted.

As regards expected inflation, it follows the same pattern as the realized inflation. It is less volatile, however, and significantly smoother than the realized series. Until the beginning of 2008 it remained close to two percent and shortly before the start of the financial crisis it picked up slightly. Following the relatively steep decline it returned again to two percent. After the outbreak of the sovereign debt crisis in the euro area expectations gradually declined until the beginning of 2015 to approximately 0.75 percent. In course of the pickup in economic activity inflation expectations gradually moved back again to slightly less than 1.7 percent at the end of 2018. With the slowdown of the economy and the subsequent outbreak of the pandemic it declined continuously until mid-2020 and has been increasing since then. The expected growth gap series follows the pattern of the realized series but exhibits a smooth dynamics. However, in general it captures periods of overheating and positive growth gap less well. Given its forward looking nature the spike in the growth gap occurs two periods earlier, as after the first wave of the pandemic with the reopening of the economy it became foreseeable that the economic activity will return to approximately previous levels.

4. Results

To estimate the model I applied Stan, a state-of-the-art, freely available software which implements the HMC algorithm in combination with the No-U-Turn (NUTS) sampler.⁶ For the baseline estimation I trained the model with the priors from Section

⁶To summarize, the NUTS sampler ensures that in the extended space, in that Hamiltionian dynamics is applied to find an update proposal, the HMC algorithm stops at some point where the distance from the original point of departure does not increase anymore. Hence, the algorithm makes no u-turn which would lead to a waste of computational resources. For further details on the NUTS sampler I refer to Hoffman and Gelman (2014).

		Pri	Prior		Posterior			
Param.	Density	Mean	Std	Mean	Std	[0.05, 0.95]		
π^*	Normal	1.90	0.15	1.87	0.11	[1.72, 2.07]		
σ	InvGamma	0.20	0.20	0.20	0.056	[0.12, 0.30]		
σ_{ν_1}	InvGamma	0.10	0.05	0.18	0.14	[0.063, 0.44]		
σ_{ν_2}	InvGamma	0.10	0.05	0.16	0.094	[0.061, 0.34]		
$\theta_{1,0}$	Normal	0.50	2.00	0.91	0.44	[0.26, 1.67]		
$ heta_{2,0}$	Normal	0.50	2.00	0.89	0.37	[0.37, 1.54]		
$\overline{ heta}_1$	_	_	_	0.90	0.55	[0.11, 1.84]		
$\overline{ heta}_2$	_	_	_	0.76	0.22	[0.41, 1.14]		

2 on the data sample from 1999.2 to 2001.4. The fine-tuning of the HMC/NUTS **Table 1:** Priors and posterior estimates – training sample from 1999.2 to 2001.4

Notes: The table shows the prior distributions set for the estimation of the model on the training sample comprising the time period from 1999.2 to 2001.4 and the posterior means, standard deviations and the 90 percent credible intervals (in brackets) of the particular parameters. The values for $\bar{\theta}_1$ and $\bar{\theta}_2$ correspond to the average of the estimated posterior means and standard deviations of each individual $\theta_{1,t}$ and $\theta_{2,t}$ at time t over the period from 1999.2 to 2001.4.

sampler was carried out automatically by Stan based on a burn-in sample of 10,000 draws for each of the four chains. The estimates for the posterior distributions were obtained based on 50,000 sample draws per chain. The estimated posterior means and standard deviations for the training sample are summarized in Table 1. These posteriors were taken then as priors for the main estimation exercise making use of the complete data set until 2021.2.

For the period from 1999.2 to 2001.4 the posterior mean for the inflation target, π^* , is estimated at 1.87 percent which corresponds almost to its prior mean. Yet, the standard deviation is lower than initially assumed implying that the data is informative. The posterior means of the initial values of both response coefficients, $\theta_{1,0}$ and $\theta_{2,0}$ are considerably higher with approximately 0.9 than their prior means and significantly more narrow with estimated standard deviations of 0.44 and 0.37, respectively. The averages of the posterior estimates of each individual $\theta_{1,t}$ and $\theta_{2,t}$ at time t over the period from 1999.2 to 2001.4, denoted by $\overline{\theta}_1$ and $\overline{\theta}_2$, are estimated at 0.90 and 0.76. This implies that the response to the expected growth gap declined over the first three years. The conditional standard deviations of the

random walk processes of the time-varying reaction coefficients, σ_{ν_1} and σ_{ν_2} , are lower with 0.18 and 0.16 than initially assumed and their posteriors are more tightly shaped. The estimation based on the complete sample was carried out again with four



Figure 2: Posterior estimates of time-varying reaction coefficients

Notes: Plot (a) shows the estimates for the time-varying response to the expected inflation gap, $\theta_{1,t}$, and plot (b) to the expected growth gap, $\theta_{2,t}$. The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate rate series, respectively. The dashed lines correspond to the estimates for the period from 2002.1 to 2021.2 and the solid lines to the period from 1999.2 to 2021.2. The priors are set according to the results in Table 1.

parallel chains, each with a burn-in of 10,000 draws to fine-tune the HMC/NUTSsampler and subsequently with 50,000 sample draws. For the sake of completeness, I carried out the estimation both on the sample from 2002.1 to 2021.2 by excluding the training sample and on all available data from 1999.2 to 2021.2 as well, see Figure 2. To this respect, the results differ only marginally. Including the data in the sample from 1999.2 to 2001.4 as well results in a smooth continuation of the estimates. However, one needs to acknowledge that for the latter case the priors are not entirely valid as also future information is used. The intention was to showcase whether and to what extent the estimation results with respect to the future response are influenced by the initial period of a seemingly aggressive response by the ECB at the end of the 1990s when reputation had to be build up. The evolution of the response to the expected inflation gap is shown in plot (a) and to the expected growth gap in plot (b). The blue lines refer to the estimation results obtained by using shadow rates from 2008.3 onwards provided by Krippner while the red lines show the results when carrying out the estimation with the Wu-Xia shadow rate series.⁷ In general, both reaction coefficients feature a declining trend over time. In particular, the average response of the ECB to expected deviations from the inflation target has declined from its relatively high initial values but stabilized then around 0.5. With the beginning of the financial crisis, the response of the ECB to the expected inflation gap has declined further though. Considering the estimation results based on the shadow rates provided by Krippner, the softening continued and became statistically significant towards the end of the sample period. By contrast, the results obtained with the Wu-Xia series suggest that the response to the expected inflation gap rebounded in the aftermath of the financial crisis and reached values in the range of 0.5. Following its local peak at the beginning of 2016 it has been declining at an increasing pace and reached almost as low values as the response obtained with the Krippner series. The difference in the parameter estimates can be explained by the steep negative progression of the Wu-Xia shadow rate series. The response to the expected growth gap has also declined since the financial crisis and converged to zero if estimated with the Krippner series. The estimation results obtained with the Wu-Xia series exhibit a similar pattern, yet the response is more volatile. After 2015 it reaches even negative values but turns out to be positive at the end of the sample period.

The posterior estimates for the time-invariant parameters and the initial values of the response to the expected inflation gap and the expected growth gap are shown in Table 2. The estimated posterior mean of the inflation target, π^* , is slightly lower than the estimate obtained using the training sample. Both estimates of π^* obtained with the two different shadow rate series, are almost similar, 1.83 if estimated with the Krippner series and 1.81 with the Wu-Xia series. Furthermore, the estimated posterior standard deviations are also lower with 0.09 and 0.08, respectively, than the estimates obtained with the training sample. The posterior mean of σ , the standard deviation of the remainder in the policy rule, is approximately twice as large if compared with the estimate based on the training sample. With the Krippner series

 $^{^7\}mathrm{Figure}$ A.1 in the Appendix shows the results in separate charts including the 90 percent credible intervals.

it is estimated at 0.43 that is larger to some extent than the result obtained with the Wu-Xia series, namely 0.37. The posterior means of the conditional standard deviations of the time-varying response to the expected inflation gap, σ_{ν_1} , are lower than for the training sample and are fairly similar for the two different shadow rates series, estimated at 0.13 and 0.14, respectively. The standard deviations of the posterior distributions of σ_{ν_1} are considerably lower with 0.080 and 0.093. The posterior mean of the conditional standard deviation of the time-varying response to the expected growth gap, σ_{ν_2} , differs though for the two different data sets, estimated at 0.10 with the Krippner series and at 0.18 with the Wu-Xia series. Hence, the Wu-Xia series generates a more volatile response to the expected growth gap than the Krippner series. Furthermore, the posterior standard deviations of σ_{ν_2} for both shadow rate series, estimated at 0.035 and 0.084 respectively, are also lower than the estimates obtained with the training sample, especially the one for the Krippner series.

		Kripp	oner	Wu-Xia			
Param.	Mean	Std	[0.05, 0.95]	Mean	Std	[0.05, 0.95]	
π^*	1.83	0.090	[1.68, 1.98]	1.81	0.080	[1.69, 1.95]	
σ	0.43	0.035	[0.38, 0.49]	0.37	0.042	[0.31, 0.44]	
σ_{ν_1}	0.13	0.080	[0.057, 0.26]	0.14	0.093	[0.059, 0.26]	
σ_{ν_2}	0.10	0.035	[0.057, 0.17]	0.18	0.084	[0.074, 0.34]	
$\theta_{1,0}$	0.66	0.38	[0.063, 1.30]	0.73	0.37	[0.14, 1.36]	
$ heta_{2,0}$	0.71	0.17	[0.43, 1.00]	0.74	0.19	[0.44, 1.06]	

 Table 2: Posterior estimates for the time period from 1999.2 to 2021.2

Notes: Posterior estimates of the time-invariant parameters and the initial values of the time-varying reaction coefficients obtained with the Krippner and the Wu-Xia shadow rate series, respectively, for the time period from 1999.2 to 2021.2. The corresponding 90 percent credible intervals are displayed in brackets. The priors are set according to the results in Table 1.

The sophisticated diagnostics unique to the HMC/NUTS algorithm did not signal any severe issues related to the sampling which would invalidate the results.⁸

⁸The Stan package comes along with a handy diagnostics interface, ShinyStan. An introduction to the diagnostics which is superior to those available to commonly applied MCMC-based algorithms, can be found in the references in the documentation available to Stan Development Team (2017).

In particular, there are no divergent sample draws at all. This indicates that the posterior likelihood surface does not exhibit any severe irregularities such as cliffs or areas with high local curvature which would prevent the sampler from exploring the posterior likelihood surface properly. In addition, the sampler did not reach the maximum tree-depth implying that the NUTS sampler could always traverse the posterior likelihood surface for an optimal distance along the Hamiltonian path. However, the energy level diagnostics, see Figure A.2 in the Appendix, suggests that the posterior exhibits heavy tails which may be challenging to sample from. Fig-

Figure 3: Posterior distribution estimates of time-invariant parameters



Notes: Posterior distribution estimates of the time-invariant parameters: inflation target, π^* , standard deviation of the error term in the policy rule, σ , and standard deviations of the random walk innovations governing the time-varying reaction coefficients, σ_{ν_1} and σ_{ν_2} . The priors are set according to the results in Table 1. Estimation period: 1999.2–2021.2.

ure 3 shows the marginal distributions of the posterior with respect to the inflation target, π^* , the standard deviation of the error term in the policy rule, σ , and the standard deviations of the innovations in the random walk process characterizing the

reaction coefficients, σ_{ν_1} and σ_{ν_2} . The plots in the left column show the marginal distributions obtained with the Krippner series and the right column those obtained with the Wu-Xia series. While the marginal distributions of π^* and σ have a regular shape, the distributions of σ_{ν_1} appear to have long tails for both shadow rates series. Furthermore, the distribution of σ_{ν_2} obtained with the Wu-Xia shadow rate series is strongly skewed. The long tails in the distributions of σ_{ν_1} indicate that there is a positive probability for the response to the expected inflation gap to be far more volatile than initially expected. Long or heavy tails may postulate an issue for the sampler as the tail of the distribution requires a relatively large step size to be efficiently explored by the HMC algorithm while to deal with the higher local curvature of the posterior density surface in the region where the probability mass is concentrated a relatively small step size may be appropriate. A relatively large step size could result in unstable dynamics so that the sampler produces divergent draws which would invalidate the results while a relatively small step size may lead to a random walk behaviour. In the optimum, when the likelihood surface does not suffer from irregularities at all, the HMC algorithm produces nearly uncorrelated sample draws due to its gradient based approach and ability to cover large distances in the parameter space. However, in this case the step size was calibrated to be relatively small due to the high curvature in the central region of the posterior. The correlogramm of the sample draws for the conditional standard deviation of the response to the expected inflation gap, σ_{ν_1} , see Figure A.3 in the Appendix, exhibits a relatively high autocorrelation of the sample draws. This explains the large amount of sample draws required to achieve an effective sample size of at least 1,000. In this case, this does not constitute a major issue though, as the trajectories of the four different chains do not feature any obvious irregularities which would prevent the sampler from exploring the posterior likelihood surface properly. There is no indication that the chains would have stuck in a certain region and the scatter plots of σ_{ν_1} , σ_{ν_2} and σ_{ν_3} do not show any obvious issues with identification either, such as a ridge, see Figure A.4 in the Appendix. They indicate, however, a slight bimodality which is more pronounced in case the Wu-Xia series is used for the estimation. However, this does not constitute an issue for the sampler either, as the energy barrier between the two regions is not as high that it would cause the sampler to stuck in one of the modes. In general, long tails do not constitute an issue, as the sampling algorithm is feasible and given the sufficiently high effective sample size, there are no irregularities which would invalidate the results.⁹

5. Robustness checks and extensions

No model training. A straightforward question to be asked is how results change depending on the priors assumed. I estimated the model on the entire data sample from 1999.2 to 2021.2 directly with the priors from Section 2 without conducting any training or subsequent adjustments in the priors. Figure 4 shows the results in comparison to the baseline estimates from Section 4 with priors based on the results in Table 1. The main difference in both estimates occurs in the estimate for the Figure 4: Time-varying response from 1999.2 to 2021.2 without model training



Notes: Plot (a) shows the estimates for the time-varying response to the expected inflation gap, $\theta_{1,t}$, and plot (b) to the expected growth gap, $\theta_{2,t}$. The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate series, respectively. The solid lines correspond to the results obtained with the priors from Section 2 and the dashed lines display the baseline estimates from Section 4 with priors according to the results in Table 1.

response to the expected inflation gap. For both shadow rate series, the response to the expected inflation gap at the initial point, $\theta_{1,0}$, is markedly lower than in the baseline estimation. The lower estimate for $\theta_{1,0}$ alters the slope of the trajectory of

⁹For the detailed Stan output containing the effective sample sizes see Table B.2 and B.3 in the Appendix.

 $\theta_{1,t}$, yet the difference to the baseline estimate gradually vanishes until 2008.4. The lower estimates until 2008.4 are most likely driven by the significantly lower prior value set for $\theta_{1,0}$, in particular 0.50 instead of 0.90. Consequently, the estimate for the response at the beginning depends significantly on the prior belief about the commitment to keep inflation close to the target at the inception of monetary union. The standard deviation of the estimation is larger though as shown in Table 3. The

	Prior				Kripp	oner	Wu-Xia			
Par.	Dens.	Mean	Std	Mean	Std	[0.05, 0.95]	Mean	Std	[0.05, 0.95]	
$ \begin{array}{c} \pi^* \\ \sigma \\ \sigma_{\nu_1} \\ \sigma_{\nu_2} \\ \theta_{1,0} \\ \theta_{2,0} \end{array} $	Norm InvG InvG InvG Norm Norm	$ \begin{array}{r} 1.9 \\ 0.2 \\ 0.2 \\ 0.2 \\ 0.5 \\ 0.5 \\ 0.5 \\ \end{array} $	$\begin{array}{c} 0.15 \\ 0.20 \\ 0.20 \\ 0.20 \\ 2.00 \\ 2.00 \end{array}$	$1.83 \\ 0.45 \\ 0.13 \\ 0.10 \\ 0.43 \\ 0.61$	$\begin{array}{c} 0.12 \\ 0.038 \\ 0.067 \\ 0.038 \\ 0.49 \\ 0.29 \end{array}$	$ \begin{bmatrix} 1.64, 2.04 \\ [0.40, 0.52] \\ [0.051, 0.23] \\ [0.05, 0.17] \\ [-0.32, 1.29] \\ [0.15, 1.09] \end{bmatrix} $	$1.81 \\ 0.39 \\ 0.13 \\ 0.18 \\ 0.56 \\ 0.67$	$\begin{array}{c} 0.11 \\ 0.045 \\ 0.078 \\ 0.092 \\ 0.51 \\ 0.41 \end{array}$	$ \begin{bmatrix} 1.66, 2.00 \\ [0.32, 0.47] \\ [0.054, 0.25] \\ [0.067, 0.36] \\ [-0.22, 1.44] \\ [0.042, 1.36] \end{bmatrix} $	

Table 3: Priors and posterior estimates without model training – 1999.2-2021.2

Notes: The posterior means and standard deviations of the time-invariant parameters as well as the initial values of the time-varying parameters were estimated on the complete data sample from 1999.2 to 2021.2 without any estimation of the model on a training sample. The corresponding 90 percent credible intervals are displayed in brackets.

estimate for $\theta_{1,0}$ obtained with the baseline setup in Section 4 lies also well within the 66 percent credible interval of this estimate. The difference between the estimates for $\theta_{1,t}$ is negligible from 2008.4 onwards. This implies that for the period after the financial crisis the data pins down the time-varying parameters fairly well. This could be attributed to the larger variation in the shadow rate series than in the MRO. The estimation with respect to the response to the expected growth gap is very small throughout the entire period for $\theta_{2,t}$. However, there is some slight difference at the very beginning of the sample if compared with the baseline estimation results which vanishes very rapidly. Similarly to $\theta_{1,0}$, this is also very likely to be attributable to the smaller prior assumed for $\theta_{2,0}$, 0.50 against 0.76 in the baseline setup. As regards the time-invariant parameters, see Table 3, the estimates for the conditional standard deviations of the random walk innovations, σ_{ν_1} and σ_{ν_2} , and for the inflation target, π^* , are almost similar to those obtained with the baseline setup from Table 2. The estimates for the standard deviation of the error term in the policy rule, σ , are slightly larger than before, 0.45 as opposed to 0.43 and 0.39 to 0.37. Therefore, the variation of the estimated time-varying response to the expected inflation and growth gap remains similar. As regards the diagnostics, there were no divergent transitions which would invalidate the results, similarly to the other estimations. However, a very large amount of sample draws is still needed to achieve an acceptably high effective sample size because of the long tails in the marginal distributions of σ_{ν_1} and σ_{ν_2} .

Less varying response. To assess how results change when less variation is permitted in the reaction coefficients I set a tighter prior distribution along with a lower expected value for the standard deviations of the innovations, σ_{ν_1} and σ_{ν_2} . I assume that each of the latter two parameters is again inverse gamma distributed, yet with an expected value of 0.1 and a standard deviation of 0.05 instead of assuming 0.2 for both the expected value and the standard deviation as in Section 2. The rest of the priors corresponds to those in Section 2 and in the previous subsection. I estimated the modified model first by following the baseline procedure, that is, I estimated it on the training sample from 1999.2 to 2001.4 and then set the obtained results as new priors to finally rerun the estimation on the complete data sample from 1999.2 to 2021.2. Afterwards I also carried out the estimation using the complete data sample without training the model on any data sample.

Figure 5 shows that the tighter prior with the lower expected value renders the reaction coefficients smoother, as expected. A further consequence of the reduced variation is that the response to the expected inflation gap is initially lower than in the baseline estimation. Yet, the difference between the curves vanishes. In case the model is estimated without training, the initial difference is more pronounced owing to the markedly smaller priors for the starting values, $\theta_{1,0}$ and $\theta_{2,0}$. The difference in the response to the expected growth gap compared to the baseline results is much smaller than in case of the expected inflation gap. The response to the expected growth gap is essentially the same with the Krippner series and is slightly larger with the Wu-Xia series in the period between 2002 and 2008, whereas at the peak in



Figure 5: Time-varying response with tight priors from 1999.2 to 2021.2

Notes: The blue and the red lines refer to the results obtained with the Krippner and the Wu-Xia shadow rate series, respectively. The solid lines show the estimates from the tight prior setup, yet obtained by training the model first on the data sample from 1999.2 to 2001.4, while the dashed lines refer to the estimates on the complete data sample from 1999.2 to 2021.2 without any model training. The dotted lines display the baseline estimates from Section 4 for comparison purposes.

2008.4 it is smaller. Afterwards the results do not differ much from those obtained with the baseline setup. The posterior mean of the inflation target ranges between 1.83 and 1.84 which is slightly higher for the Wu-Xia series while it remains essentially the same for the Krippner series, see Table B.4 and B.5 in the Appendix. The volatility of the deviation from the policy rule is very slightly higher if compared with the results from the baseline setup. It is estimated at 0.44 and 0.46 for the Krippner series and at 0.40 and 0.42 for the Wu-Xia series, compared to 0.43 and 0.37 from Table 2. This suggests that the volatility in the data translates into higher deviations from the policy rule. In general, one can conclude that the main difference lies in the estimates for the response to the expected inflation gap in the initial period. Again, no issues were signaled by the diagnostics. The effective sample sizes of σ_{ν_1} and σ_{ν_2} improved slightly owing to the shorter tail in the marginal distribution, however compared with the rest of the parameters it is still very low.

Correlated response. A further canonical extension of the model is to allow for a cross correlation between the error terms of the time-varying reaction coefficients. Hence, I release the assumption that Q is diagonal. I will maintain though the assumption that the initial states, θ_0 , are independently distributed and also independent from the rest of the priors. Therefore, I can assume that the prior distribution of Q is Inverse-Wishart with ~ $\mathcal{W}^{-1}(\text{diag}(0.18^2, 0.16^2), 4)$. Thereby, I leave Q relatively unrestricted and let the data decide about the cross-covariances. For the rest of the priors the results from Table 1 are used. Figure 6 shows that allowing for cross-correlation changes the evolution of the response to both the expected inflation gap and growth gap over time only very slightly, compared to the baseline results. Yet, the estimates suggest a slight positive cross correlation between the innovations in the reaction coefficients with 0.17 and 0.08 for the Krippner and the Wu-Xia series, respectively. Using the priors from Section 2 for the rest of the parameters results in an even lower cross correlation of 0.10 and 0.05 for the two datasets, respectively, while results remain almost similar to the case where no model training is carried out. The corresponding results are shown in Figure A.6 in the Appendix.





Notes: The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate series, respectively. The solid lines correspond to the results obtained by allowing for correlation between the innovations in the random walk processes governing the reaction coefficients. The dashed lines display the baseline estimates from Section 4 for comparison purposes.

Stochastic volatility. To account for periods of both higher and lower variation in the deviation from the policy rule I extend the model by assuming stochastic volatility for the error term in the policy rule, so I postulate the following structure for the disturbance in the observation equation:

$$\epsilon_t := \gamma_t e^{\frac{s_t}{2}}, \ \gamma_t \sim N(0, 1).$$
(19)

The volatility process, s_t can be described as follows:

$$s_t = \mu_s + \rho_s (s_{t-1} - \mu_s) + \psi_t \tau_s$$
(20)

$$\psi_t \sim N(0,1), \quad s_0 \sim N\left(\mu_s, \frac{\tau_s^2}{1-\rho_s^2}\right)$$

$$(21)$$

where ψ_t is the shock to the volatility and τ_s the corresponding scaling parameter. Combining the equations, the conditional distribution of s_t is given by

$$(s_t|s_{t-1},\mu_s,\rho_s,\tau_s) \sim N\left(\mu_s + \rho_s(s_{t-1}-\mu_s),\tau_s^2\right).$$
 (22)

The conditional distribution of ϵ_t equals to:

$$(\epsilon_t | s^T, \mu_s, \rho_s, \tau_s) = (\epsilon_t | s_t, \mu_s, \rho_s, \tau_s) \sim N(0, v_t^2)$$
(23)

with $v_t := e^{\frac{s_t}{2}}$. Since I assume that there is no cross correlation between the error terms of the state space, the likelihood translates to

$$p(Y^T | \tilde{\theta}^T, \sigma_{\nu_1}, \sigma_{\nu_2}, \mu_s, \rho_s, \tau_s, s^T) = \prod_{t=0}^T \frac{1}{(2\pi v_t^2)^{1/2}} \left[\exp\left\{ -\frac{(y_t - f(X_t, \tilde{\theta}_t))^2}{2v_t^2} \right\} \right].$$
(24)

In addition, I assume that all parameters determining s_t are independently distributed from the rest of the parameters. As a consequence, the joint prior distribution can be written as

$$p(\tilde{\theta}^{T}, \sigma_{\nu_{1}}, \sigma_{\nu_{2}}, \mu_{s}, \rho_{s}, \tau_{s}, s^{T}) = p(\theta^{T} | \pi^{*}, \sigma_{\nu_{1}}, \sigma_{\nu_{2}}, \mu_{s}, \rho_{s}, \tau_{s}, s^{T}) p(\pi^{*}, \sigma_{\nu_{1}}, \sigma_{\nu_{2}}, \mu_{s}, \rho_{s}, \tau_{s}, s^{T}) = p(\pi^{*}) p(\sigma_{\nu_{1}}) p(\sigma_{\nu_{2}}) p(\mu_{s}, \rho_{s}, \tau_{s}) p(s^{T} | \mu_{s}, \rho_{s}, \tau_{s}) p(\theta_{0}) \prod_{t=0}^{T-1} p(\theta_{t+1} | \theta_{t}, \sigma_{\nu_{1}}, \sigma_{\nu_{2}}) = p(\pi^{*}) p(\sigma_{\nu_{1}}) p(\sigma_{\nu_{2}}) p(\mu_{s}) p(\rho_{s}) p(\tau_{s}) p(s_{0}) p(\theta_{0}) \prod_{t=0}^{T-1} p(s_{t+1} | s_{t}, \mu_{s}, \rho_{s}, \tau_{s}) \prod_{t=0}^{T-1} p(\theta_{t+1} | \theta_{t}, \sigma_{\nu_{1}}, \sigma_{\nu_{2}}) = (25)$$

To estimate the model I assume that ρ_s is uniformly distributed in the interval (-1, 1)and both μ_s and τ_s are Cauchy distributed with $C(0, 5)^{10}$. Figure 7 shows the results in comparison to those from the baseline estimation. The first main difference is that

¹⁰The probability density function of the Cauchy distribution is given by $p(x; a, b) = \frac{1}{\pi b \left[1 + \left(\frac{x-a}{b}\right)^2\right]}$.



Figure 7: Time-varying response with stochastic volatility from 1999.2 to 2021.2

Notes: The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate rate series, respectively. The solid lines show the estimates of the time-varying reaction coefficients assuming stochastic volatility in the deviations from the policy rule. The dashed lines display the baseline estimates from Section 4 for comparison purposes.

until the beginning of the financial crisis the response to the expected inflation gap is far more volatile than without modelling stochastic volatility. It decreases until end-2004 and increases then sharply until mid-2007. Although inflation hovered around the target until the financial crisis, the response was considerably weaker for a relatively short period than initially suggested by the baseline model. This finding is also supported by the response to the expected growth gap which decreases as well until end-2004 and increases sharply afterwards until the outbreak of the financial crisis. Given that the policy rate remained constant between 2003.3 and 2005.4 these results are hardly surprising. After the beginning of the financial crisis the response to the expected inflation gap depends strongly on the data series used. The estimation based on the Krippner series exhibits the same pattern as in the baseline estimation. It decreases continuously until the end of the sample period where it reaches again negative values. In contrast, the results obtained with the Wu-Xia series indicate that after a relatively moderate decline the response strengthens and reaches a new peak. However, afterwards it decreases continuously and ends up at a relatively low level, yet not zero. The pattern of the response to the expected growth gap following the start of the financial crisis is comparable with that from the baseline estimation irrespective of the data series used. For the Krippner series initially it decreases but rebounds then relatively fast and peaks in the second half of 2012, yet drops to zero in the aftermath. For the Wu-Xia series it declines smoothly and drops to roughly zero towards the end of the sample.



Figure 8: Posterior estimates of time-varying volatility from 1999.2 to 2021.2

Notes: The blue and the red lines refer to the evolution of the of the volatility in the deviations from the policy rule obtained with the Krippner and with the Wu-Xia shadow rate rate series, respectively.

The estimated evolution of the volatility, see Figure 8, suggests that until the beginning of the financial crisis the deviation from the estimated response tended to be smaller and less volatile. Allowing for a varying volatility in the deviations from the policy rule renders the reaction coefficient smaller which provides a better fit. In contrast, with a constant volatility parameter the larger deviations from the policy rule following the financial crisis increase the volatility of the deviations uniformly also for the period before the financial crisis. This results in a more forceful estimated response along with a larger deviation from the policy rule whereas in reality the response might have been less forceful and the deviations from it small. Therefore, it is essential to account for changes in the volatility over time.

In the aftermath of the financial crisis the magnitude of the deviations increased remarkably as with non-standard measures there is obviously less room for a fine tuning of the policy rate. These results imply that the central bank may have been more aggressive for quiet some time after the financial crisis than initially estimated without assuming changes in the volatility in the deviations from the policy rule over time. Since the central bank does not have the ability to set policy rates directly, deviations from the assumed response are larger. Therefore, for the agents in the economy it will be more difficult to distinguish between an actual change in the systematic response and a larger deviation from the policy rule. This may render policy less efficient, as agents may build expectations relying on inaccurate values with respect to the strength of the response. A further notable result is that although technically s_t was restricted to be stationary, there might be little need to allow for permanent changes in the magnitude of the deviations from the policy rule as ρ_s is estimated at 0.75 and 0.66 for the Krippner and the Wu-Xia series, respectively. This implies that deviations from the intended policy response occur only on a temporary basis, typically when the economy is hit by large shocks, for example at the beginning of the financial crisis or at the outbreak of the recent pandemic.

The diagnostics did not signal any serious issues with sampling. However, the estimation of the model with the Krippner series resulted in one single diverging sample draw within overall 200,000 sample draws. This means that in a single case when the new proposal draw was calculated by applying the numerical solution algorithm to the Hamiltonian equation it drifted off the Hamiltonian path. Usually this occurs when the local curvature of the joint posterior is very large for the calibrated step size. However, in case of this more sophisticated model a single divergent sample draw can be ignored as the relative amount of divergent sample draws is marginal if compared with the overall number sample draws. Moreover, the one divergent sample draw does not exhibit any pattern that would indicate that the algorithm is unstable in a certain region and therefore the sampler cannot access certain regions of the joint posterior. To remedy this issue it is recommended to reduce the step size of the HMC algorithm. However, to achieve this result the step size had to be reduced further manually by an immense magnitude compared to the automatically calibrated value by the software package used. This indicates, that the expansion of the model with stochastic volatility results in a way more complex joint posterior surface which is more complicated to be sampled from and increases the runtime immensely.¹¹ Using common non-gradient based sampling methods, as the random

¹¹Without modelling stochastic volatility the runtime is typically less than 5 minutes on a commonly available desktop computer equipped with an AMD 3950x CPU. The expansion of the model with stochastic volatility increased the runtime to 2-3 days in order to obtain 200,000 sample draws with only one divergent draw. With the Wu-Xia series there were no divergences among the 200,000 sample draws. However, the step size had to be reduced approximately by the same magnitude and resulted in a very similar runtime as for the estimation with the Krippner series.

walk Metropolis Hastings algorithm, it is questionable whether sampling would be tractable at all. Furthermore, without sophisticated diagnostic features unique to the HMC algorithm it could easily occur that one would not even realize that the sampler could not access certain parts of the posterior and end up with biased results.

New inflation target. In course of its strategy review in 2021 the ECB has announced its new symmetric inflation target of 2 percent. Therefore, I assume that from 2021.3 onwards π^* equals 2 + d percent, where d is assumed to be normally distributed with $d \sim N(0, 0.1^2)$. Given the recent inflation developments and to achieve a better identification I truncate d at zero and exclude thereby negative values. Redefining $\tilde{\theta}^T$ by adding d, the arguments from Section 2 hold, so the joint posterior can be expanded by the prior distribution of d. I estimated the model first without assuming stochastic volatility. Figure 9 shows that the response to the expected





Notes: Plot (a) shows the estimates for the time-varying response to the expected inflation gap, $\theta_{1,t}$, and plot (b) to the expected growth gap, $\theta_{2,t}$. The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate rate series, respectively. The solid lines refer to the estimates using priors based on Table 1. The dashed lines refer to the estimates using priors from Section 2. The dashed lines refer to the estimates using priors from Section 2 with the tight prior setup for the standard deviations of the innovation in the time-varying parameters.

inflation gap is considerably influenced by the recent response to its sharp rise. For the Krippner series, the estimates in the period until the start of the financial crisis depend again on which initial priors are assumed as in the baseline estimation until 2021.2. However, if compared with the results from the baseline setup, after 2017 the reaction function does not decrease anymore, but increases again except

a short interruption between 2019.4 and 2020.1. At the end of the sample period the response to the expected inflation gap has reached its highest value since 2007.3. The fact that towards the end of the sample period the estimates are influenced substantially by the most recent data is inherent in the random walk structure of the reaction coefficients. In periods when the time-varying parameters are less well identified or the relationship with the dependent variable might be weak, the values are influenced considerably by the parameter values in those periods with a better identification and a stronger relationship with the dependent variable. For the Wu-Xia series the first main difference is that the model prefers a more volatile response to the expected inflation gap over time. Moreover, the strong response at the end of the sample period influences the values significantly from approximately 2012 onwards. Following the decrease before the outbreak of the pandemic the response to the expected inflation gap increases sharply and reaches even more than two times higher values than before the start of the financial crisis. Setting even tighter priors for the conditional standard deviation of the reaction coefficients does not alter the main pattern, yet renders the evolution smoother. As regards the response with respect to the expected growth gap the evolution is very similar if compared with the baseline results using data only until 2021.2. Afterwards the response increases slightly, yet it still remains in relatively low ranges.

In light of the recent rise in both realized and expected inflation a natural question to be asked is the whether recently the ECB has become more tolerant against inflation and to what extend the true inflation target has changed. For the period until 2021.2 the inflation target with the Krippner series is estimated at 1.82 being essentially similar to the results from the baseline setup. Using the Wu-Xia series the target is estimated to be 1.74 percent which is somewhat lower than the baseline estimate. As regards the short period following the strategy review, the parameter d is estimated at 0.08 and 0.06 for the Krippner and the Wu-Xia series, respectively. This suggests that the inflation target is estimated at very slightly above the symmetric 2 percent target.

The model was also reestimated with the stochastic volatility setup from above.



Figure 10: Time-varying response with stochastic volatility from 1999.2 to 2022.3

Notes: Posterior estimates of the time-varying reaction coefficients assuming stochastic volatility in the deviations from the policy rule. The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate rate series, respectively.

Figure 10 suggests that with the Krippner series the response to the expected inflation gap exhibits a continuously falling tendency from the beginning of 2014 and reaches the trough only in 2020.1, while without stochastic volatility it appears to have been increasing gradually since 2017. In addition, from 2020.2 the increase is steeper and ends up at 0.48 at the end of the sample period being higher than the estimated value of 0.38 without modelling stochastic volatility. For the Wu-Xia series, the stochastic volatility setup renders the evolution of the response to the inflation gap relatively smooth. Moreover, the increase from 2020.2 is far less steep than without assuming stochastic volatility. It is also considerably lower even if compared with the case from Figure 9 where tight priors are used for the standard deviation of the innovations to prevent the time-varying reaction process from sudden fluctuations. Nevertheless, the response reaches its highest value ever at the end of the sample period. As regards the response to the expected growth gap, using the Krippner series it is slightly more volatile which results in turn in a slightly higher response towards the end of the sample period than without modelling stochastic volatility. For the Wu-Xia series the response is far less volatile than without stochastic volatility and notably lower at the end of the sample period.

Finally, the estimation results for the volatility suggest that for the Krippner series the deviations from the policy rule have fallen back to levels prevailing before the financial crisis following the immense pandemic shock. The results with respect

Figure 11: Posterior estimates of time-varying volatility from 1999.2 to 2022.3



Notes: The blue and the red lines refer to the evolution of the of the volatility in the deviations from the policy rule obtained with the Krippner and with the Wu-Xia shadow rate rate series, respectively.

to the Wu-Xia series suggest though that the volatility has kept rising and reached its highest levels ever. As regards the persistence of the changes in the magnitude of the volatility in the deviations from the policy rule, for the Krippner series ρ_s is slightly lower with 0.73 than the estimate without considering the post strategic review period. In contrast, for the Wu-Xia series it increases from 0.66 to 0.80. The latter result suggests, that deviations from the policy rule may have become recently longer lasting. Thus, monetary policy may remain more aggressive than the policy rule would suggest. In the future this could translate in a higher estimated systematic response.

6. Conclusion

In order to assess the changes in the monetary policy stance of the euro area since its inception I estimated a monetary policy function, which has been shown to adequately describe the ECB's policy response (see e.g. OW, BW and HS). To that end, I developed a Bayesian time-varying parameter framework and sampled from the joint posterior distribution of the parameters by using the Hamiltonian Monte Carlo algorithm. I also improved on existing studies by relying on two different shadow interest-rate series instead of key policy rates. That permitted to estimate the impact of non-conventional measures on the ECB's reaction function directly during periods when monetary policy may have been constrained by the effective lower bound. In addition, this flexible framework allowed for a more sophisticated modelling of the stochastic volatility associated with deviations from the policy rule.

I found that the ECB's response to the expected inflation gap and the expected growth gap has varied considerably over time. While the response to the expected growth gap has weakened substantially in the aftermath of the financial crisis, the estimation results for the response to the expected inflation gap are somewhat mixed and depend on the shadow rate series used for estimation purposes. In particular, the majority of the results suggests that the response weakened since the onset of the financial crisis and when quantitative measures were in place. There are also indications, however, that this weakening in the response to expected inflation was less pronounced.

If the period following the ECB's strategy review is also included into the data sample used for the estimation, the results imply that the ECB's response has recently become more forceful. However, while the results obtained with Krippner's shadow rate series suggest that the response to the expected inflation gap remains well below its historical peak, the results obtained by using the alternative shadow rate series provided by Wu and Xia show that it recently achieved its highest level ever. The estimation results obtained with the extended data sample including the post-strategy review period also indicate that the response to the expected inflation gap weakened much less during the years prior to the pandemic, especially if the model is estimated with the Wu-Xia shadow rate series. This feature is inherent in the model because periods of weak responses, or which may be less well identified by the data, are strongly affected by subsequent peaks in the response. This may make the estimated policy response more forceful during periods when the response was indeed weak. I also found that the inflation target rose to slightly above 2 percent as a result of the ECB's recent strategy review. Consequently, the results do not suggest much greater tolerance of a large deviation from the new target.

Factoring stochastic volatility into deviations from the policy rule over time is essential as it strongly influences the estimation results. Firstly, results suggest that the policy response in the run-up to the financial crisis was more volatile than that suggested by the baseline estimation setup. It declined significantly from its initially high levels until 2005 but then rebounded sharply. Secondly, the response to the expected inflation gap after the financial crisis was more forceful than it would have been without modelling stochastic volatility. Thirdly, the response to the expected inflation gap declined more sharply after quantitative easing was introduced in 2015.1. And, lastly, the recent increase in the response to the sharp rise in the expected inflation gap is much smaller, especially when the Wu-Xia shadow rate series is used for estimation, while with the Krippner series it is only slightly higher.

Given the importance of both the time variability of the response parameters and the stochastic volatility in the deviations from the policy rule, it is essential to take account of both when investigating possible changes in the stance of monetary policy. The HMC algorithm is a suitable approach for dealing with extremely complex and very high-dimensional models. Consequently, extensions of the present model in terms of its statistical and structural properties are easily possible to improve accuracy and structural interpretation. I leave these topics for future research.

References

- **Betancourt, Michael**, "A Conceptual Introduction to Hamiltonian Monte Carlo," 2017.
- Blattner, Tobias Sebastian and Emil Margaritov, "Towards a robust monetary policy rule for the euro area," Working Paper Series 1210, European Central Bank 2010.
- Bletzinger, Tilman and Volker Wieland, "Lower for longer: The case of the ECB," *Economics Letters*, 2017, 159 (C), 123–127.
- Boivin, Jean, "Has U.S. Monetary Policy Changed? Evidence from Drifting Coefficients and Real-Time Data," *Journal of Money, Credit and Banking*, August 2006, 38 (5), 1149–1173.
- Clarida, Richard, Jordi Gali, and Mark Gertler, "Monetary policy rules and macroeconomic stability: evidence and some theory," *The Quarterly Journal of Economics*, 2000, 115 (1), 147–180.
- Cogley, Timothy and Thomas J Sargent, "Evolving post-world war II US inflation dynamics," *NBER macroeconomics annual*, 2001, *16*, 331–373.
- and _ , "Drifts and volatilities: monetary policies and outcomes in the post
 WWII US," *Review of Economic Dynamics*, 2005, 8 (2), 262–302.
- Gerlach, Stefan and John Lewis, "ECB Reaction Functions and the Crisis of 2008," International Journal of Central Banking, 2014, 10 (1), 137–158.
- and _ , "Zero lower bound, ECB interest rate policy and the financial crisis," Empirical Economics, May 2014, 46 (3), 865–886.
- Gorter, Janko, Fauve Stolwijk, Jan PAM Jacobs, and Jakob de Haan, "ECB policy making and the financial crisis," Working Paper Series 272, De Nederlandsche Bank 2010.

- Hartmann, Philipp and Frank Smets, "The European Central Bank's monetary policy during its first 20 years," *Brookings Papers on Economic Activity*, 2018, 2018 (2), 1–146.
- Hoffman, Matthew D and Andrew Gelman, "The No-U-Turn sampler: adaptively setting path lengths in Hamiltonian Monte Carlo.," J. Mach. Learn. Res., 2014, 15 (1), 1593–1623.
- Judd, John P, Glenn D Rudebusch et al., "Taylor's Rule and the Fed: 1970-1997," Economic Review - Federal Reserve Bank of San Francisco, 1998, pp. 3–16.
- Kim, Chang-Jin and Charles R Nelson, "Estimation of a forward-looking monetary policy rule: A time-varying parameter model using ex post data," *Journal* of Monetary Economics, 2006, 53 (8), 1949–1966.
- Koop, Gary, Bayesian econometrics, John Wiley & Sons, 2003.
- Krippner, Leo, "Measuring the stance of monetary policy in zero lower bound environments," *Economics Letters*, 2013, 118 (1), 135–138.
- _, Zero lower bound term structure modeling: A practitioner's guide, Springer, 2015.
- Maih, Junior, Falk Mazelis, Roberto Motto, and Annukka Ristiniemi, "Asymmetric monetary policy rules for the euro area and the US," *Journal of Macroeconomics*, 2021, 70 (C).
- Neal, Radford M, "MCMC using Hamiltonian dynamics," in Steve Brooks, Andrew Gelman, Galin L. Jones, and Xiao-Li Meng, eds., Handbook of Markov Chain Monte Carlo, Chapman & Hall/CRC, 2011, chapter 5, pp. 113–162.
- **Orphanides, Athanasios**, "Monetary policy rules, macroeconomic stability, and inflation: A view from the trenches," *Journal of Money, Credit and Banking*, 2004, *36* (2), 151–175.
- and Volker Wieland, "Complexity and Monetary Policy," International Journal of Central Banking, 2013, 9 (1), 167–204.

- Paloviita, Maritta, Markus Haavio, Pirkka Jalasjoki, and Juha Kilponen, "What Does "Below, but Close to, 2 Percent" Mean? Assessing the ECB's Reaction Function with Real-Time Data," *International Journal of Central Banking*, June 2021, 17 (2), 125–169.
- Rostagno, Massimo, Carlo Altavilla, Giacomo Carboni, Wolfgang Lemke, Roberto Motto, Arthur Saint Guilhem, and Jonathan Yiangou, "A tale of two decades: the ECB's monetary policy at 20," Working Paper Series 2346, European Central Bank 2019.
- Sims, Christopher A, "Stability and instability in US monetary policy behavior," Unpublished Paper, 2001.
- and Tao Zha, "Were there regime switches in US monetary policy?," American Economic Review, 2006, 96 (1), 54–81.
- Smets, Frank, "Comment on Chapters 6 and 7," in Marco Buti, Servaas Deroose, Vitor Gaspar, and Joao Nogueira Martins, eds., *The Euro: the First Decade*, Cambridge University Press, 2010, pp. 259–280.
- Stan Development Team, "shinystan: Interactive Visual and Numerical Diagnostics and Posterior Analysis for Bayesian Models.," 2017. R package version 2.4.0.
- Taylor, John B., "Discretion versus policy rules in practice," in "Carnegie-Rochester conference series on public policy," Vol. 39 Elsevier 1993, pp. 195–214.
- Wu, Jing Cynthia and Fan Dora Xia, "Time-varying lower bound of interest rates in Europe," *Chicago Booth Research Paper*, 2017, (17-06).
- and _ , "Negative interest rate policy and the yield curve," Journal of Applied Econometrics, 2020, 35 (6), 653–672.

Appendix A. Figures



Figure A.1: Posterior estimates of time-varying reaction coefficients

Krippner





Notes: Plot (a) and (b): estimation results obtained with the Krippner shadow rate series. Plot (c) and (d): estimation results obtained with the Wu-Xia shadow rate series. The solid lines refer to the estimation on the complete data sample from 1999.2 to 2021.2 and dashed lines to the sample from 2002.1 to 2021.2. The grey shaded area and the dotted lines show the 90 percent credible intervals with respect to the estimation from 1999.2 to 2021.2 and from 2002.1 to 2021.2, respectively.





Notes: The figure shows the overlaid histograms of the (centered) marginal energy distribution and the first-differenced distribution which is referred to as the energy diagnostics. Plot (a) displays the diagnostics of each chain obtained with the Krippner shadow rate series while plot (b) shows the diagnostics of each chain obtained with the Wu-Xia shadow rate series. The estimation was carried out with data covering the period from 1999.2 to 2021.2.



Figure A.3: Correlogramms of time-invariant parameter sample draws

Notes: The plot shows correlogramms of the sample draws for the time-invariant parameters. The left and the right column contain the results obtained with the Krippner shadow rate series and with the Wu-Xia shadow rate series, respectively. Estimation period: 1999.2-2021.2.



Figure A.4: Scatter plots of sample draws of standard deviations

Notes: Sample draws of the standard deviations of the random walk innovations, σ_{ν_1} and σ_{ν_2} , and the standard deviation of the error term in the policy rule, σ .



Figure A.5: Posterior distributions of time-invariant parameters – 2002.1-2021.2

Notes: Posterior distributions estimates of the time-invariant parameters: inflation target, π^* , standard deviation of the error term in the policy rule policy rule, σ , standard deviations of the innovations in the processes governing the time-varying reaction coefficients, σ_{ν_1} and σ_{ν_2} . Priors are set according to the results from Table 1. Estimation period: 2002.1–2021.2.

Figure A.6: Correlated time-varying response without model training from 1999.2 to 2021.2



Notes: The blue and the red lines refer to the results obtained with the Krippner and with the Wu-Xia shadow rate series, respectively. The solid lines correspond to the results obtained by allowing for correlation between the innovations of the random walk processes governing the reaction coefficients without any model training. The dashed lines display the baseline setup without any model training from Section 5 for comparison purposes.

Appendix B. Tables

			Kripp	oner		Wu-Xia				
Param.	Density	Mean	Std	[0.05, 0.95]	Mean	Std	[0.05, 0.95]			
π^*	Normal	1.82	0.098	[1.66, 1.98]	1.80	0.088	[1.67, 1.95]			
σ	InvGamma	0.45	0.038	[0.39, 0.52]	0.39	0.046	[0.31, 0.46]			
σ_{ν_1}	InvGamma	0.12	0.070	[0.057, 0.25]	0.13	0.082	[0.058, 0.26]			
σ_{ν_2}	InvGamma	0.11	0.039	[0.059, 0.18]	0.20	0.097	[0.078, 0.38]			
$\theta_{1,0}$	Normal	0.52	0.40	[-0.11, 1.21]	0.59	0.40	[-0.058, 1.26]			
$ heta_{2,0}$	Normal	0.61	0.18	[0.31, 0.92]	0.62	0.19	[0.30, 0.94]			

Table B.1: Posterior estimates for the time period from 2002.1 to 2021.2

Notes: Posterior estimates of the time-invariant parameters and the initial values of the timevarying reaction coefficients for the period from 2002.1 to 2021.2. Priors are set according to the results in Table 1.

Param.	Rhat	\mathbf{N}_{eff}	Mean	Std	2.5%	25%	50%	75%	97.5%
π^*	1.0000	177129	1.8268	0.0903	1.6572	1.7660	1.8234	1.8843	2.0142
σ	1.0001	32882	0.4331	0.0352	0.3682	0.4093	0.4317	0.4554	0.5063
$\sigma_{ u_1}$	1.0010	2504	0.1268	0.0796	0.0511	0.0816	0.1071	0.1461	0.3239
$\sigma_{ u_2}$	1.0005	6173	0.1010	0.0350	0.0523	0.0762	0.0946	0.1185	0.1868
$\theta_{1,1999.2}$	1.0001	37739	0.6567	0.3770	-0.0449	0.3973	0.6437	0.9015	1.4329
$\theta_{1,1999.3}$	1.0001	52780	0.6358	0.3809	-0.0738	0.3759	0.6218	0.8807	1.4205
$\theta_{1,1999.4}$	1.0001	64063	0.6181	0.3885	-0.1054	0.3565	0.6032	0.8651	1.4207
$\theta_{1,2000.1}$	1.0001	61360	0.6114	0.3992	-0.1276	0.3449	0.5949	0.8603	1.4394
$\theta_{1,2000.2}$	1.0001	60528	0.6025	0.4088	-0.1533	0.3323	0.5859	0.8543	1.4501
$\theta_{1,2000.3}$	1.0002	62000	0.5936	0.4157	-0.1740	0.3214	0.5760	0.8458	1.4596
$\theta_{1,2000.4}$	1.0001	62700	0.5842	0.4209	-0.1939	0.3105	0.5655	0.8379	1.4632
$\theta_{1,2001.1}$	1.0001	61544	0.5755	0.4240	-0.2066	0.3008	0.5557	0.8305	1.4625
$\theta_{1,2001.2}$	1.0001	60632	0.5653	0.4291	-0.2273	0.2888	0.5460	0.8205	1.4632
$\theta_{1,2001.3}$	1.0001	63907	0.5541	0.4325	-0.2455	0.2762	0.5355	0.8106	1.4511
$\theta_{1,2001.4}$	1.0001	62250	0.5447	0.4330	-0.2533	0.2668	0.5256	0.8019	1.4459
$\theta_{1,2002.1}$	1.0001	74195	0.5253	0.4337	-0.2833	0.2493	0.5096	0.7847	1.4237
$\theta_{1,2002.2}$	1.0001	78044	0.5099	0.4338	-0.3064	0.2347	0.4957	0.7705	1.4042
$\theta_{1,2002.3}$	1.0001	78458	0.4957	0.4339	-0.3243	0.2232	0.4828	0.7570	1.3874
$\theta_{1,2002.4}$	1.0001	78093	0.4809	0.4329	-0.3454	0.2102	0.4691	0.7424	1.3638

Table B.2: Posterior estimates – Krippner shadow rate series

$\theta_{1,2003.1}$	1.0001	79540	0.4670	0.4288	-0.3577	0.1976	0.4575	0.7277	1.3407
$\theta_{1,2003.2}$	1.0001	75129	0.4517	0.4241	-0.3705	0.1861	0.4441	0.7114	1.3099
$\theta_{1,2003.3}$	1.0001	77333	0.4431	0.4190	-0.3726	0.1790	0.4362	0.7015	1.2924
$\theta_{1,2003.4}$	1.0001	74488	0.4327	0.4211	-0.3865	0.1712	0.4257	0.6905	1.2848
$\theta_{1,2004.1}$	1.0001	76178	0.4275	0.4209	-0.3926	0.1655	0.4208	0.6837	1.2813
$\theta_{1,2004.2}$	1.0001	79934	0.4238	0.4211	-0.3909	0.1632	0.4156	0.6797	1.2788
$\theta_{1,2004.3}$	1.0001	79640	0.4205	0.4208	-0.3920	0.1609	0.4119	0.6741	1.2730
$\theta_{1,2004.4}$	1.0001	80903	0.4172	0.4190	-0.3900	0.1584	0.4089	0.6691	1.2666
$\theta_{1,2005.1}$	1.0001	81104	0.4142	0.4158	-0.3862	0.1573	0.4054	0.6633	1.2579
$\theta_{1,2005.2}$	1.0001	80050	0.4131	0.4120	-0.3753	0.1578	0.4036	0.6594	1.2539
$\theta_{1,2005.3}$	1.0001	80023	0.4115	0.4065	-0.3648	0.1582	0.4016	0.6537	1.2435
$\theta_{1,2005.4}$	1.0001	74776	0.4113	0.3990	-0.3479	0.1606	0.4011	0.6505	1.2323
$\theta_{1,2006.1}$	1.0001	59257	0.4119	0.3903	-0.3313	0.1645	0.3991	0.6464	1.2180
$\theta_{1,2006.2}$	1.0001	46531	0.4113	0.3793	-0.3078	0.1690	0.3988	0.6403	1.2036
$\theta_{1,2006.3}$	1.0001	31865	0.4136	0.3774	-0.2964	0.1729	0.3977	0.6369	1.2067
$\theta_{1,2006.4}$	1.0001	30288	0.4052	0.3725	-0.2903	0.1687	0.3899	0.6248	1.1869
$\theta_{1,2007.1}$	1.0001	28268	0.3944	0.3685	-0.2916	0.1607	0.3789	0.6106	1.1673
$\theta_{1,2007.2}$	1.0001	30300	0.3786	0.3625	-0.2962	0.1487	0.3647	0.5899	1.1351
$\theta_{1,2007.3}$	1.0001	33753	0.3607	0.3544	-0.3012	0.1376	0.3480	0.5673	1.0939
$\theta_{1,2007.4}$	1.0001	40023	0.3406	0.3429	-0.3018	0.1231	0.3305	0.5437	1.0462
$\theta_{1,2008.1}$	1.0000	43653	0.3215	0.3315	-0.3075	0.1107	0.3138	0.5204	0.9970
$\theta_{1,2008.2}$	1.0000	49818	0.3017	0.3163	-0.3034	0.0986	0.2958	0.4955	0.9426
$\theta_{1,2008.3}$	1.0000	74209	0.2749	0.2996	-0.3111	0.0809	0.2717	0.4650	0.8765
$\theta_{1,2008.4}$	1.0001	60526	0.2228	0.3026	-0.3910	0.0365	0.2292	0.4176	0.8004
$\theta_{1,2009.1}$	1.0001	30721	0.1916	0.3038	-0.4338	0.0108	0.2026	0.3879	0.7550
$\theta_{1,2009.2}$	1.0002	14661	0.1525	0.3099	-0.4973	-0.0236	0.1696	0.3523	0.7038
$\theta_{1,2009.3}$	1.0001	23283	0.1562	0.2897	-0.4526	-0.0197	0.1683	0.3474	0.6934
$\theta_{1,2009.4}$	1.0000	69869	0.1779	0.2780	-0.3903	0.0018	0.1840	0.3608	0.7109
$\theta_{1,2010.1}$	1.0000	90803	0.1902	0.2823	-0.3824	0.0117	0.1944	0.3727	0.7372
$\theta_{1,2010.2}$	1.0000	94548	0.1930	0.2908	-0.3935	0.0112	0.1963	0.3781	0.7576
$\theta_{1,2010.3}$	1.0000	97785	0.1824	0.2992	-0.4319	-0.0011	0.1888	0.3733	0.7579
$\theta_{1,2010.4}$	1.0000	73422	0.1757	0.3113	-0.4689	-0.0103	0.1838	0.3729	0.7676
$\theta_{1,2011.1}$	1.0000	79970	0.1809	0.3225	-0.4835	-0.0085	0.1891	0.3813	0.7949
$\theta_{1,2011.2}$	1.0000	87274	0.1867	0.3310	-0.4915	-0.0054	0.1941	0.3892	0.8181
$\theta_{1,2011.3}$	1.0000	78210	0.1901	0.3393	-0.5057	-0.0029	0.1981	0.3955	0.8362
$\theta_{1,2011.4}$	1.0000	109455	0.2070	0.3392	-0.4754	0.0096	0.2106	0.4084	0.8724

$\theta_{1,2012.1}$	1.0000	106654	0.2187	0.3427	-0.4585	0.0175	0.2187	0.4187	0.9008
$\theta_{1,2012.2}$	1.0000	105884	0.2215	0.3430	-0.4536	0.0191	0.2205	0.4202	0.9077
$\theta_{1,2012.3}$	1.0000	106166	0.2245	0.3402	-0.4426	0.0227	0.2225	0.4231	0.9054
$\theta_{1,2012.4}$	1.0000	103971	0.2246	0.3347	-0.4323	0.0245	0.2215	0.4219	0.8982
$\theta_{1,2013.1}$	1.0000	104175	0.2269	0.3267	-0.4115	0.0300	0.2228	0.4200	0.8862
$\theta_{1,2013.2}$	1.0000	64795	0.2382	0.3181	-0.3740	0.0422	0.2308	0.4258	0.8939
$\theta_{1,2013.3}$	1.0000	99362	0.2249	0.3065	-0.3761	0.0356	0.2218	0.4103	0.8438
$\theta_{1,2013.4}$	1.0000	38492	0.2475	0.3037	-0.3245	0.0553	0.2376	0.4256	0.8811
$\theta_{1,2014.1}$	1.0000	31247	0.2510	0.2957	-0.3042	0.0628	0.2406	0.4250	0.8719
$\theta_{1,2014.2}$	1.0000	34851	0.2435	0.2840	-0.2943	0.0608	0.2348	0.4142	0.8370
$\theta_{1,2014.3}$	1.0001	32649	0.2338	0.2743	-0.2852	0.0569	0.2253	0.4002	0.8031
$\theta_{1,2014.4}$	1.0001	78084	0.1988	0.2581	-0.3058	0.0313	0.1954	0.3631	0.7210
$\theta_{1,2015.1}$	1.0001	93561	0.1588	0.2508	-0.3439	-0.0030	0.1601	0.3224	0.6513
$\theta_{1,2015.2}$	1.0001	40345	0.1314	0.2662	-0.4187	-0.0303	0.1369	0.3028	0.6402
$\theta_{1,2015.3}$	1.0000	37740	0.2003	0.2735	-0.3194	0.0258	0.1926	0.3656	0.7635
$\theta_{1,2015.4}$	1.0001	17871	0.2316	0.2843	-0.2871	0.0493	0.2171	0.3955	0.8384
$\theta_{1,2016.1}$	1.0001	19929	0.2230	0.2806	-0.2947	0.0412	0.2099	0.3886	0.8188
$\theta_{1,2016.2}$	1.0000	62638	0.1779	0.2650	-0.3328	0.0048	0.1730	0.3436	0.7209
$\theta_{1,2016.3}$	1.0000	62593	0.1532	0.2658	-0.3642	-0.0192	0.1493	0.3202	0.6926
$\theta_{1,2016.4}$	1.0001	22540	0.0689	0.2850	-0.5286	-0.1004	0.0802	0.2524	0.5987
$\theta_{1,2017.1}$	1.0001	25549	0.0560	0.2951	-0.5666	-0.1180	0.0683	0.2455	0.6035
$\theta_{1,2017.2}$	1.0000	40306	0.0612	0.2995	-0.5660	-0.1157	0.0712	0.2519	0.6240
$\theta_{1,2017.3}$	1.0000	87358	0.0764	0.3020	-0.5419	-0.1063	0.0809	0.2663	0.6583
$\theta_{1,2017.4}$	1.0000	101482	0.0838	0.3069	-0.5338	-0.1029	0.0866	0.2745	0.6809
$\theta_{1,2018.1}$	1.0000	101244	0.0811	0.3128	-0.5507	-0.1068	0.0838	0.2747	0.6900
$\theta_{1,2018.2}$	1.0000	89478	0.1056	0.3143	-0.5062	-0.0878	0.1016	0.2936	0.7369
$\theta_{1,2018.3}$	1.0000	44096	0.1237	0.3190	-0.4802	-0.0743	0.1161	0.3098	0.7772
$\theta_{1,2018.4}$	1.0001	30090	0.1356	0.3216	-0.4602	-0.0644	0.1245	0.3180	0.8056
$\theta_{1,2019.1}$	1.0001	16835	0.1519	0.3266	-0.4333	-0.0512	0.1355	0.3302	0.8418
$\theta_{1,2019.2}$	1.0002	13141	0.1568	0.3274	-0.4210	-0.0470	0.1377	0.3331	0.8526
$\theta_{1,2019.3}$	1.0002	11100	0.1526	0.3255	-0.4160	-0.0492	0.1339	0.3251	0.8355
$\theta_{1,2019.4}$	1.0000	96088	0.0465	0.2818	-0.5212	-0.1323	0.0503	0.2279	0.5969
$\theta_{1,2020.1}$	1.0001	30377	0.0008	0.2916	-0.5994	-0.1781	0.0091	0.1903	0.5537
$\theta_{1,2020.2}$	1.0000	54261	-0.0017	0.2911	-0.5963	-0.1855	0.0041	0.1881	0.5577
$\theta_{1,2020.3}$	1.0000	64106	-0.0084	0.2962	-0.6144	-0.1961	-0.0035	0.1859	0.5647
$\theta_{1,2020.4}$	1.0000	51970	-0.0228	0.3219	-0.6900	-0.2217	-0.0137	0.1874	0.5855

$\theta_{1,2021.1}$	1.0001	29772	-0.0465	0.3480	-0.7848	-0.2548	-0.0317	0.1808	0.6008
$\theta_{1,2021.2}$	1.0001	31254	-0.0510	0.3743	-0.8482	-0.2701	-0.0340	0.1903	0.6377
$\theta_{2,1999.2}$	1.0000	146229	0.7111	0.1741	0.3743	0.5933	0.7097	0.8269	1.0582
$\theta_{2,1999.3}$	1.0000	144693	0.7035	0.1817	0.3528	0.5813	0.7009	0.8230	1.0681
$\theta_{2,1999.4}$	1.0000	138541	0.6956	0.1832	0.3418	0.5728	0.6930	0.8151	1.0648
$\theta_{2,2000.1}$	1.0000	158879	0.6777	0.1820	0.3247	0.5564	0.6755	0.7965	1.0424
$\theta_{2,2000.2}$	1.0000	164822	0.6638	0.1798	0.3142	0.5438	0.6625	0.7820	1.0204
$\theta_{2,2000.3}$	1.0000	168282	0.6497	0.1802	0.2980	0.5300	0.6488	0.7685	1.0079
$\theta_{2,2000.4}$	1.0000	165101	0.6430	0.1867	0.2778	0.5198	0.6420	0.7654	1.0143
$\theta_{2,2001.1}$	1.0000	162927	0.6334	0.1932	0.2549	0.5068	0.6322	0.7593	1.0182
$\theta_{2,2001.2}$	1.0000	158466	0.6258	0.1965	0.2400	0.4972	0.6249	0.7530	1.0175
$\theta_{2,2001.3}$	1.0000	156512	0.6176	0.1953	0.2346	0.4894	0.6169	0.7442	1.0069
$\theta_{2,2001.4}$	1.0000	146542	0.6173	0.1921	0.2434	0.4899	0.6151	0.7428	1.0023
$\theta_{2,2002.1}$	1.0000	146512	0.5857	0.2006	0.1898	0.4541	0.5855	0.7169	0.9832
$\theta_{2,2002.2}$	1.0000	138855	0.5660	0.2100	0.1453	0.4301	0.5675	0.7030	0.9772
$\theta_{2,2002.3}$	1.0000	127120	0.5470	0.2160	0.1095	0.4091	0.5499	0.6877	0.9683
$\theta_{2,2002.4}$	1.0000	116133	0.5307	0.2193	0.0855	0.3902	0.5339	0.6742	0.9567
$\theta_{2,2003.1}$	1.0000	115133	0.5222	0.2225	0.0713	0.3795	0.5250	0.6685	0.9555
$\theta_{2,2003.2}$	1.0001	102986	0.5067	0.2267	0.0456	0.3619	0.5104	0.6557	0.9449
$\theta_{2,2003.3}$	1.0001	98950	0.4999	0.2333	0.0264	0.3514	0.5042	0.6527	0.9514
$\theta_{2,2003.4}$	1.0000	97741	0.4920	0.2447	-0.0084	0.3380	0.4969	0.6514	0.9655
$\theta_{2,2004.1}$	1.0000	98318	0.4885	0.2546	-0.0338	0.3304	0.4929	0.6528	0.9819
$\theta_{2,2004.2}$	1.0000	103340	0.4852	0.2617	-0.0535	0.3243	0.4900	0.6533	0.9921
$\theta_{2,2004.3}$	1.0001	105101	0.4819	0.2676	-0.0694	0.3169	0.4869	0.6528	1.0021
$\theta_{2,2004.4}$	1.0000	103642	0.4787	0.2718	-0.0772	0.3119	0.4833	0.6514	1.0081
$\theta_{2,2005.1}$	1.0000	103703	0.4750	0.2746	-0.0864	0.3062	0.4796	0.6482	1.0107
$\theta_{2,2005.2}$	1.0000	104344	0.4717	0.2756	-0.0911	0.3029	0.4748	0.6452	1.0113
$\theta_{2,2005.3}$	1.0000	106964	0.4685	0.2753	-0.0910	0.2995	0.4709	0.6417	1.0125
$\theta_{2,2005.4}$	1.0000	107753	0.4665	0.2738	-0.0911	0.2982	0.4681	0.6386	1.0071
$\theta_{2,2006.1}$	1.0000	110993	0.4643	0.2710	-0.0849	0.2976	0.4653	0.6338	1.0019
$\theta_{2,2006.2}$	1.0000	113264	0.4625	0.2670	-0.0771	0.2978	0.4629	0.6292	0.9951
$\theta_{2,2006.3}$	1.0000	113697	0.4616	0.2628	-0.0665	0.2992	0.4608	0.6245	0.9876
$\theta_{2,2006.4}$	1.0000	116186	0.4602	0.2567	-0.0528	0.3006	0.4595	0.6195	0.9744
$\theta_{2,2007.1}$	1.0000	119459	0.4585	0.2486	-0.0355	0.3031	0.4569	0.6128	0.9597
$\theta_{2,2007.2}$	1.0000	120184	0.4592	0.2392	-0.0153	0.3084	0.4570	0.6074	0.9430
$\theta_{2,2007.3}$	1.0000	114385	0.4601	0.2276	0.0139	0.3162	0.4567	0.6004	0.9237

$\theta_{2,2007.4}$	1.0000	106835	0.4594	0.2142	0.0436	0.3225	0.4547	0.5920	0.8984
$\theta_{2,2008.1}$	1.0000	96075	0.4585	0.1979	0.0777	0.3308	0.4535	0.5813	0.8656
$\theta_{2,2008.2}$	1.0000	75313	0.4589	0.1784	0.1195	0.3415	0.4537	0.5706	0.8287
$\theta_{2,2008.3}$	1.0000	50576	0.4677	0.1596	0.1706	0.3604	0.4613	0.5677	0.8004
$\theta_{2,2008.4}$	1.0000	23777	0.4939	0.1468	0.2272	0.3944	0.4852	0.5847	0.8068
$\theta_{2,2009.1}$	1.0000	60638	0.4027	0.1153	0.1800	0.3254	0.4014	0.4787	0.6339
$\theta_{2,2009.2}$	1.0001	38197	0.2898	0.1214	0.0439	0.2105	0.2920	0.3715	0.5226
$\theta_{2,2009.3}$	1.0001	42880	0.2631	0.1450	-0.0388	0.1721	0.2683	0.3602	0.5349
$\theta_{2,2009.4}$	1.0000	60897	0.2549	0.1622	-0.0843	0.1543	0.2603	0.3628	0.5598
$\theta_{2,2010.1}$	1.0000	74563	0.2476	0.1750	-0.1173	0.1389	0.2533	0.3632	0.5782
$\theta_{2,2010.2}$	1.0001	90128	0.2448	0.1829	-0.1324	0.1302	0.2498	0.3647	0.5939
$\theta_{2,2010.3}$	1.0000	120277	0.2507	0.1904	-0.1376	0.1302	0.2539	0.3751	0.6180
$\theta_{2,2010.4}$	1.0000	136965	0.2537	0.1962	-0.1428	0.1287	0.2555	0.3812	0.6379
$\theta_{2,2011.1}$	1.0000	135338	0.2464	0.2010	-0.1565	0.1180	0.2478	0.3773	0.6399
$\theta_{2,2011.2}$	1.0000	132815	0.2359	0.2035	-0.1722	0.1046	0.2381	0.3682	0.6341
$\theta_{2,2011.3}$	1.0000	125380	0.2196	0.2066	-0.1977	0.0876	0.2225	0.3544	0.6226
$\theta_{2,2011.4}$	1.0000	138222	0.2326	0.2108	-0.1885	0.0966	0.2329	0.3696	0.6511
$\theta_{2,2012.1}$	1.0000	126169	0.2427	0.2132	-0.1761	0.1040	0.2414	0.3803	0.6703
$\theta_{2,2012.2}$	1.0000	119926	0.2352	0.2187	-0.1934	0.0928	0.2337	0.3754	0.6758
$\theta_{2,2012.3}$	1.0000	117566	0.2221	0.2223	-0.2160	0.0778	0.2214	0.3649	0.6668
$\theta_{2,2012.4}$	1.0000	115157	0.2017	0.2282	-0.2516	0.0551	0.2020	0.3485	0.6550
$\theta_{2,2013.1}$	1.0000	109163	0.1768	0.2330	-0.2902	0.0274	0.1784	0.3280	0.6342
$\theta_{2,2013.2}$	1.0000	98911	0.1672	0.2376	-0.3115	0.0154	0.1691	0.3208	0.6338
$\theta_{2,2013.3}$	1.0000	94712	0.1548	0.2397	-0.3273	0.0018	0.1564	0.3098	0.6257
$\theta_{2,2013.4}$	1.0000	82154	0.1296	0.2410	-0.3603	-0.0233	0.1335	0.2869	0.5971
$\theta_{2,2014.1}$	1.0000	74585	0.1178	0.2411	-0.3739	-0.0346	0.1216	0.2750	0.5846
$\theta_{2,2014.2}$	1.0000	75964	0.1134	0.2422	-0.3773	-0.0398	0.1173	0.2699	0.5849
$\theta_{2,2014.3}$	1.0000	77504	0.1124	0.2434	-0.3819	-0.0414	0.1160	0.2697	0.5861
$\theta_{2,2014.4}$	1.0000	84143	0.1247	0.2435	-0.3630	-0.0306	0.1261	0.2813	0.6037
$\theta_{2,2015.1}$	1.0000	84761	0.1394	0.2413	-0.3364	-0.0159	0.1377	0.2930	0.6227
$\theta_{2,2015.2}$	1.0000	80200	0.1480	0.2354	-0.3115	-0.0054	0.1452	0.2979	0.6231
$\theta_{2,2015.3}$	1.0000	77781	0.0929	0.2327	-0.3760	-0.0560	0.0957	0.2445	0.5476
$\theta_{2,2015.4}$	1.0001	63972	0.0673	0.2363	-0.4161	-0.0819	0.0723	0.2218	0.5222
$\theta_{2,2016.1}$	1.0001	70787	0.0712	0.2391	-0.4166	-0.0795	0.0757	0.2268	0.5361
$\theta_{2,2016.2}$	1.0000	88653	0.0977	0.2402	-0.3839	-0.0546	0.0992	0.2509	0.5743
$\theta_{2,2016.3}$	1.0000	95125	0.1174	0.2413	-0.3590	-0.0366	0.1163	0.2691	0.6058

$\theta_{2,2016.4}$	1.0000	83374	0.1500	0.2415	-0.3129	-0.0071	0.1441	0.2995	0.6516
$\theta_{2,2017.1}$	1.0000	72081	0.1622	0.2377	-0.2885	0.0068	0.1543	0.3085	0.6606
$\theta_{2,2017.2}$	1.0000	71811	0.1642	0.2310	-0.2736	0.0129	0.1566	0.3070	0.6496
$\theta_{2,2017.3}$	1.0000	74840	0.1575	0.2223	-0.2658	0.0118	0.1509	0.2954	0.6208
$\theta_{2,2017.4}$	1.0000	73433	0.1567	0.2146	-0.2513	0.0152	0.1502	0.2897	0.6052
$\theta_{2,2018.1}$	1.0000	58324	0.1652	0.2105	-0.2298	0.0252	0.1569	0.2949	0.6091
$\theta_{2,2018.2}$	1.0000	83011	0.1358	0.2037	-0.2516	0.0023	0.1296	0.2630	0.5598
$\theta_{2,2018.3}$	1.0000	102143	0.1160	0.2025	-0.2735	-0.0145	0.1102	0.2414	0.5367
$\theta_{2,2018.4}$	1.0000	116612	0.1029	0.2001	-0.2804	-0.0259	0.0980	0.2258	0.5210
$\theta_{2,2019.1}$	1.0000	133434	0.0865	0.1958	-0.2915	-0.0381	0.0817	0.2055	0.4917
$\theta_{2,2019.2}$	1.0000	150393	0.0695	0.1872	-0.2949	-0.0485	0.0655	0.1831	0.4560
$\theta_{2,2019.3}$	1.0000	178870	0.0484	0.1746	-0.2970	-0.0602	0.0463	0.1554	0.4034
$\theta_{2,2019.4}$	1.0000	171428	0.0146	0.1569	-0.3054	-0.0811	0.0161	0.1129	0.3231
$\theta_{2,2020.1}$	1.0000	156466	0.0009	0.1312	-0.2680	-0.0795	0.0033	0.0842	0.2567
$\theta_{2,2020.2}$	1.0000	152337	-0.0009	0.0879	-0.1759	-0.0583	-0.0005	0.0567	0.1724
$\theta_{2,2020.3}$	1.0000	232644	-0.0061	0.0858	-0.1795	-0.0597	-0.0054	0.0482	0.1628
$\theta_{2,2020.4}$	1.0000	151244	-0.0138	0.0390	-0.0908	-0.0398	-0.0136	0.0124	0.0625
$\theta_{2,2021.1}$	1.0000	71996	0.0398	0.0925	-0.1336	-0.0211	0.0362	0.0968	0.2347
$\theta_{2,2021.2}$	1.0000	82664	0.0547	0.1004	-0.1382	-0.0122	0.0528	0.1196	0.2587

Notes: Posterior estimates of the time-invariant and the time-varying parameters obtained with the shadow rate series of Krippner for the period from 1999.2 to 2021.2. Priors are set according to the results in Table 1. $\theta_{1,1999.2}$ and $\theta_{2,1999.2}$ correspond to $\theta_{1,0}$ and $\theta_{2,0}$, respectively. N_{eff} stands for the effective sample size. Calculations carried out with ShinyStan.

Param.	Rhat	N_{eff}	Mean	Std	2.5%	25%	50%	75%	97.5%
π^*	1.0000	106351	1.8134	0.0809	1.6646	1.7594	1.8088	1.8625	1.9871
σ	1.0008	7909	0.3739	0.0417	0.2925	0.3472	0.3747	0.4015	0.4536
$\sigma_{ u_1}$	1.0021	2429	0.1352	0.0928	0.0522	0.0862	0.1153	0.1569	0.3213
$\sigma_{ u_2}$	1.0013	3646	0.1779	0.0841	0.0661	0.1125	0.1598	0.2279	0.3768
$\theta_{1,1999.2}$	1.0001	37181	0.7320	0.3736	0.0250	0.4759	0.7225	0.9769	1.4921
$\theta_{1,1999.3}$	1.0001	48261	0.7089	0.3790	-0.0061	0.4513	0.6998	0.9555	1.4823
$\theta_{1,1999.4}$	1.0001	45917	0.6906	0.3897	-0.0459	0.4297	0.6812	0.9418	1.4823
$\theta_{1,2000.1}$	1.0001	50524	0.6879	0.4016	-0.0685	0.4209	0.6769	0.9429	1.5082
$\theta_{1,2000.2}$	1.0001	50198	0.6805	0.4145	-0.0957	0.4078	0.6682	0.9400	1.5257
$\theta_{1,2000.3}$	1.0000	49685	0.6733	0.4240	-0.1174	0.3969	0.6600	0.9352	1.5415

Table B.3: Posterior estimates – Wu-Xia shadow rate series

$\theta_{1,2000.4}$	1.0000	48523	0.6666	0.4285	-0.1337	0.3880	0.6525	0.9304	1.5487
$\theta_{1,2001.1}$	1.0000	45267	0.6613	0.4310	-0.1419	0.3786	0.6454	0.9249	1.5540
$\theta_{1,2001.2}$	1.0001	42741	0.6539	0.4386	-0.1634	0.3694	0.6364	0.9199	1.5586
$\theta_{1,2001.3}$	1.0001	43181	0.6443	0.4448	-0.1818	0.3588	0.6274	0.9110	1.5616
$\theta_{1,2001.4}$	1.0001	36498	0.6378	0.4488	-0.1950	0.3506	0.6203	0.9042	1.5632
$\theta_{1,2002.1}$	1.0001	47023	0.6144	0.4494	-0.2322	0.3312	0.6016	0.8847	1.5353
$\theta_{1,2002.2}$	1.0001	47843	0.5993	0.4501	-0.2565	0.3180	0.5885	0.8725	1.5139
$\theta_{1,2002.3}$	1.0001	45666	0.5865	0.4491	-0.2724	0.3048	0.5771	0.8587	1.4984
$\theta_{1,2002.4}$	1.0001	44535	0.5731	0.4498	-0.2907	0.2932	0.5654	0.8454	1.4782
$\theta_{1,2003.1}$	1.0001	43988	0.5622	0.4471	-0.3010	0.2838	0.5546	0.8342	1.4621
$\theta_{1,2003.2}$	1.0001	37291	0.5452	0.4410	-0.3203	0.2704	0.5408	0.8168	1.4313
$\theta_{1,2003.3}$	1.0001	38693	0.5391	0.4345	-0.3109	0.2659	0.5333	0.8084	1.4168
$\theta_{1,2003.4}$	1.0001	32654	0.5269	0.4361	-0.3354	0.2559	0.5236	0.7981	1.3973
$\theta_{1,2004.1}$	1.0001	33639	0.5236	0.4348	-0.3382	0.2532	0.5201	0.7932	1.3934
$\theta_{1,2004.2}$	1.0001	33920	0.5215	0.4343	-0.3353	0.2530	0.5179	0.7888	1.3929
$\theta_{1,2004.3}$	1.0001	35824	0.5205	0.4342	-0.3309	0.2537	0.5174	0.7863	1.3922
$\theta_{1,2004.4}$	1.0001	36681	0.5204	0.4319	-0.3223	0.2539	0.5165	0.7838	1.3885
$\theta_{1,2005.1}$	1.0001	37412	0.5201	0.4257	-0.3064	0.2558	0.5155	0.7806	1.3812
$\theta_{1,2005.2}$	1.0002	36843	0.5205	0.4214	-0.2972	0.2598	0.5154	0.7770	1.3725
$\theta_{1,2005.3}$	1.0001	36379	0.5217	0.4164	-0.2895	0.2643	0.5162	0.7754	1.3627
$\theta_{1,2005.4}$	1.0001	36617	0.5255	0.4078	-0.2606	0.2707	0.5192	0.7744	1.3545
$\theta_{1,2006.1}$	1.0002	34089	0.5308	0.3979	-0.2313	0.2782	0.5233	0.7737	1.3475
$\theta_{1,2006.2}$	1.0002	29449	0.5334	0.3846	-0.2040	0.2868	0.5242	0.7695	1.3232
$\theta_{1,2006.3}$	1.0003	17832	0.5433	0.3877	-0.1808	0.2955	0.5298	0.7738	1.3510
$\theta_{1,2006.4}$	1.0003	17602	0.5364	0.3830	-0.1760	0.2910	0.5225	0.7628	1.3350
$\theta_{1,2007.1}$	1.0003	15568	0.5279	0.3825	-0.1742	0.2852	0.5136	0.7497	1.3200
$\theta_{1,2007.2}$	1.0003	15090	0.5131	0.3819	-0.1853	0.2733	0.4977	0.7301	1.2963
$\theta_{1,2007.3}$	1.0003	17072	0.4942	0.3744	-0.1920	0.2595	0.4801	0.7085	1.2588
$\theta_{1,2007.4}$	1.0002	17975	0.4743	0.3648	-0.1952	0.2452	0.4619	0.6850	1.2153
$\theta_{1,2008.1}$	1.0003	14931	0.4586	0.3594	-0.1955	0.2328	0.4463	0.6630	1.1805
$\theta_{1,2008.2}$	1.0004	12644	0.4417	0.3506	-0.1946	0.2227	0.4283	0.6405	1.1440
$\theta_{1,2008.3}$	1.0004	13502	0.4120	0.3312	-0.1996	0.2012	0.4015	0.6070	1.0791
$\theta_{1,2008.4}$	1.0002	25583	0.3254	0.3211	-0.3149	0.1276	0.3318	0.5313	0.9347
$\theta_{1,2009.1}$	1.0003	10997	0.2661	0.3386	-0.4133	0.0753	0.2830	0.4813	0.8630
$\theta_{1,2009.2}$	1.0005	6794	0.2056	0.3626	-0.5274	0.0179	0.2325	0.4318	0.8034
$\theta_{1,2009.3}$	1.0002	15755	0.2281	0.3027	-0.4043	0.0379	0.2384	0.4313	0.7950

$\theta_{1,2009.4}$	1.0003	10922	0.1981	0.3124	-0.4673	0.0088	0.2129	0.4069	0.7678
$\theta_{1,2010.1}$	1.0001	22022	0.2328	0.3013	-0.3894	0.0432	0.2415	0.4320	0.8035
$\theta_{1,2010.2}$	1.0001	26573	0.2440	0.3082	-0.3920	0.0517	0.2528	0.4452	0.8267
$\theta_{1,2010.3}$	1.0002	22773	0.2391	0.3197	-0.4376	0.0458	0.2519	0.4480	0.8312
$\theta_{1,2010.4}$	1.0001	34419	0.2581	0.3244	-0.4209	0.0621	0.2690	0.4679	0.8669
$\theta_{1,2011.1}$	1.0000	40373	0.2820	0.3372	-0.4045	0.0802	0.2892	0.4917	0.9174
$\theta_{1,2011.2}$	1.0001	31532	0.3040	0.3489	-0.3913	0.0963	0.3071	0.5124	0.9621
$\theta_{1,2011.3}$	1.0001	27986	0.3217	0.3566	-0.3761	0.1119	0.3219	0.5302	0.9990
$\theta_{1,2011.4}$	1.0002	13599	0.3514	0.3758	-0.3405	0.1344	0.3429	0.5532	1.0617
$\theta_{1,2012.1}$	1.0001	21397	0.3515	0.3677	-0.3490	0.1359	0.3456	0.5579	1.0610
$\theta_{1,2012.2}$	1.0001	34389	0.3487	0.3656	-0.3581	0.1337	0.3460	0.5583	1.0484
$\theta_{1,2012.3}$	1.0000	75479	0.3431	0.3559	-0.3687	0.1334	0.3451	0.5555	1.0310
$\theta_{1,2012.4}$	1.0001	61872	0.3378	0.3532	-0.3766	0.1319	0.3438	0.5528	1.0155
$\theta_{1,2013.1}$	1.0002	25032	0.3317	0.3541	-0.3892	0.1303	0.3412	0.5488	0.9949
$\theta_{1,2013.2}$	1.0004	14770	0.3307	0.3552	-0.3892	0.1348	0.3425	0.5486	0.9838
$\theta_{1,2013.3}$	1.0001	44546	0.3663	0.3278	-0.2923	0.1661	0.3672	0.5713	1.0125
$\theta_{1,2013.4}$	1.0000	66205	0.3996	0.3190	-0.2244	0.1989	0.3960	0.5961	1.0478
$\theta_{1,2014.1}$	1.0000	59247	0.4343	0.3122	-0.1600	0.2331	0.4251	0.6252	1.0815
$\theta_{1,2014.2}$	1.0001	47350	0.4563	0.3052	-0.1198	0.2578	0.4460	0.6434	1.0924
$\theta_{1,2014.3}$	1.0001	31929	0.4831	0.2988	-0.0724	0.2851	0.4697	0.6658	1.1146
$\theta_{1,2014.4}$	1.0001	30395	0.4898	0.2871	-0.0464	0.2978	0.4775	0.6691	1.0950
$\theta_{1,2015.1}$	1.0001	38312	0.4775	0.2787	-0.0486	0.2917	0.4679	0.6535	1.0548
$\theta_{1,2015.2}$	1.0001	43864	0.4333	0.2937	-0.1371	0.2472	0.4299	0.6167	1.0233
$\theta_{1,2015.3}$	1.0001	51282	0.4496	0.3030	-0.1266	0.2565	0.4427	0.6370	1.0707
$\theta_{1,2015.4}$	1.0001	36582	0.4791	0.3042	-0.0896	0.2800	0.4664	0.6645	1.1204
$\theta_{1,2016.1}$	1.0002	21982	0.4981	0.3062	-0.0616	0.2955	0.4812	0.6810	1.1519
$\theta_{1,2016.2}$	1.0001	40801	0.4691	0.2917	-0.0822	0.2761	0.4590	0.6509	1.0746
$\theta_{1,2016.3}$	1.0001	29681	0.4757	0.2907	-0.0637	0.2817	0.4636	0.6565	1.0877
$\theta_{1,2016.4}$	1.0001	39819	0.4378	0.2952	-0.1282	0.2471	0.4308	0.6221	1.0403
$\theta_{1,2017.1}$	1.0004	14305	0.4786	0.3246	-0.0962	0.2668	0.4572	0.6632	1.1754
$\theta_{1,2017.2}$	1.0001	43224	0.4433	0.3207	-0.1497	0.2347	0.4284	0.6354	1.1227
$\theta_{1,2017.3}$	1.0000	52662	0.4113	0.3277	-0.2086	0.2046	0.4013	0.6093	1.0859
$\theta_{1,2017.4}$	1.0001	47454	0.3944	0.3375	-0.2419	0.1867	0.3860	0.5957	1.0807
$\theta_{1,2018.1}$	1.0001	54230	0.4002	0.3408	-0.2311	0.1864	0.3881	0.5997	1.1079
$\theta_{1,2018.2}$	1.0003	17909	0.4212	0.3567	-0.2023	0.1965	0.3987	0.6145	1.1782
$\theta_{1,2018.3}$	1.0003	30061	0.4007	0.3416	-0.2208	0.1824	0.3828	0.5959	1.1278

$\theta_{1,2018.4}$	1.0001	45912	0.3778	0.3323	-0.2394	0.1654	0.3639	0.5736	1.0763
$\theta_{1,2019.1}$	1.0002	39211	0.3660	0.3226	-0.2333	0.1567	0.3529	0.5584	1.0458
$\theta_{1,2019.2}$	1.0004	17481	0.3596	0.3250	-0.2305	0.1491	0.3425	0.5473	1.0487
$\theta_{1,2019.3}$	1.0005	16138	0.3312	0.3155	-0.2446	0.1258	0.3156	0.5169	0.9950
$\theta_{1,2019.4}$	1.0001	23672	0.2276	0.3032	-0.3753	0.0357	0.2278	0.4223	0.8267
$\theta_{1,2020.1}$	1.0001	33699	0.2047	0.3050	-0.3962	0.0058	0.2030	0.4031	0.8126
$\theta_{1,2020.2}$	1.0001	25593	0.1667	0.3246	-0.4759	-0.0418	0.1673	0.3765	0.8065
$\theta_{1,2020.3}$	1.0001	19504	0.1295	0.3362	-0.5474	-0.0871	0.1332	0.3493	0.7865
$\theta_{1,2020.4}$	1.0003	11003	0.0755	0.3890	-0.7410	-0.1539	0.0932	0.3275	0.7922
$\theta_{1,2021.1}$	1.0006	6212	0.0072	0.4638	-0.9920	-0.2312	0.0464	0.3003	0.7824
$\theta_{1,2021.2}$	1.0004	8614	0.0225	0.4687	-0.9988	-0.2283	0.0591	0.3207	0.8290
$\theta_{2,1999.2}$	1.0000	134485	0.7416	0.1886	0.3774	0.6135	0.7399	0.8672	1.1172
$\theta_{2,1999.3}$	1.0000	104600	0.7423	0.2276	0.3098	0.5934	0.7359	0.8838	1.2135
$\theta_{2,1999.4}$	1.0000	75155	0.7389	0.2299	0.3040	0.5876	0.7309	0.8812	1.2208
$\theta_{2,2000.1}$	1.0000	211196	0.6920	0.2208	0.2572	0.5490	0.6901	0.8334	1.1362
$\theta_{2,2000.2}$	1.0000	217235	0.6655	0.2088	0.2505	0.5293	0.6660	0.8019	1.0773
$\theta_{2,2000.3}$	1.0000	171424	0.6380	0.2061	0.2235	0.5043	0.6404	0.7740	1.0399
$\theta_{2,2000.4}$	1.0000	210876	0.6406	0.2252	0.1880	0.4969	0.6415	0.7860	1.0869
$\theta_{2,2001.1}$	1.0000	204512	0.6301	0.2493	0.1229	0.4755	0.6320	0.7870	1.1226
$\theta_{2,2001.2}$	1.0000	202050	0.6283	0.2567	0.1088	0.4707	0.6295	0.7870	1.1399
$\theta_{2,2001.3}$	1.0000	196171	0.6237	0.2397	0.1441	0.4722	0.6243	0.7765	1.0996
$\theta_{2,2001.4}$	1.0000	98299	0.6543	0.2148	0.2376	0.5113	0.6508	0.7943	1.0862
$\theta_{2,2002.1}$	1.0000	91897	0.5614	0.2460	0.0510	0.4079	0.5685	0.7218	1.0323
$\theta_{2,2002.2}$	1.0000	64017	0.5224	0.2752	-0.0651	0.3579	0.5345	0.7018	1.0359
$\theta_{2,2002.3}$	1.0001	50366	0.4912	0.2837	-0.1155	0.3210	0.5056	0.6776	1.0174
$\theta_{2,2002.4}$	1.0001	42247	0.4666	0.2792	-0.1286	0.2940	0.4817	0.6532	0.9839
$\theta_{2,2003.1}$	1.0000	60263	0.4732	0.2748	-0.0997	0.3007	0.4836	0.6560	0.9917
$\theta_{2,2003.2}$	1.0001	35141	0.4372	0.2782	-0.1496	0.2620	0.4494	0.6250	0.9537
$\theta_{2,2003.3}$	1.0001	37836	0.4336	0.2922	-0.1821	0.2510	0.4481	0.6306	0.9748
$\theta_{2,2003.4}$	1.0001	32966	0.4151	0.3336	-0.3126	0.2172	0.4375	0.6360	1.0218
$\theta_{2,2004.1}$	1.0001	37623	0.4148	0.3688	-0.4009	0.2044	0.4407	0.6542	1.0835
$\theta_{2,2004.2}$	1.0001	40650	0.4162	0.3939	-0.4571	0.1962	0.4446	0.6654	1.1375
$\theta_{2,2004.3}$	1.0001	42209	0.4178	0.4116	-0.4988	0.1901	0.4466	0.6753	1.1715
$\theta_{2,2004.4}$	1.0001	43831	0.4186	0.4254	-0.5339	0.1865	0.4489	0.6824	1.2006
$\theta_{2,2005.1}$	1.0001	43769	0.4202	0.4331	-0.5479	0.1839	0.4516	0.6876	1.2183
$\theta_{2,2005.2}$	1.0001	43576	0.4215	0.4354	-0.5520	0.1848	0.4527	0.6889	1.2275

$\theta_{2,2005.3}$	1.0001	43789	0.4228	0.4331	-0.5452	0.1851	0.4542	0.6897	1.2247
$\theta_{2,2005.4}$	1.0001	45819	0.4282	0.4305	-0.5338	0.1919	0.4575	0.6940	1.2282
$\theta_{2,2006.1}$	1.0001	47740	0.4357	0.4242	-0.5144	0.2019	0.4634	0.6967	1.2287
$\theta_{2,2006.2}$	1.0001	48959	0.4427	0.4123	-0.4746	0.2154	0.4680	0.6985	1.2162
$\theta_{2,2006.3}$	1.0001	54323	0.4544	0.4099	-0.4556	0.2299	0.4775	0.7052	1.2296
$\theta_{2,2006.4}$	1.0000	58506	0.4652	0.4023	-0.4241	0.2458	0.4861	0.7095	1.2278
$\theta_{2,2007.1}$	1.0000	65032	0.4763	0.3894	-0.3855	0.2628	0.4949	0.7120	1.2170
$\theta_{2,2007.2}$	1.0000	75788	0.4945	0.3726	-0.3226	0.2889	0.5070	0.7195	1.2121
$\theta_{2,2007.3}$	1.0000	87931	0.5133	0.3482	-0.2393	0.3181	0.5225	0.7220	1.1913
$\theta_{2,2007.4}$	1.0001	97061	0.5279	0.3224	-0.1560	0.3437	0.5334	0.7208	1.1651
$\theta_{2,2008.1}$	1.0001	97130	0.5416	0.2859	-0.0554	0.3745	0.5433	0.7150	1.1131
$\theta_{2,2008.2}$	1.0001	86381	0.5590	0.2421	0.0674	0.4110	0.5579	0.7089	1.0473
$\theta_{2,2008.3}$	1.0002	46471	0.6105	0.2068	0.2197	0.4758	0.6022	0.7373	1.0459
$\theta_{2,2008.4}$	1.0011	5826	0.7914	0.2362	0.4055	0.6173	0.7616	0.9433	1.3100
$\theta_{2,2009.1}$	1.0002	34070	0.5395	0.1207	0.3090	0.4586	0.5374	0.6175	0.7832
$\theta_{2,2009.2}$	1.0005	6969	0.2784	0.1631	-0.0575	0.1703	0.2883	0.3927	0.5711
$\theta_{2,2009.3}$	1.0001	51153	0.4426	0.1761	0.1099	0.3278	0.4366	0.5498	0.8153
$\theta_{2,2009.4}$	1.0002	22363	0.5062	0.2364	0.0998	0.3515	0.4827	0.6328	1.0571
$\theta_{2,2010.1}$	1.0001	56559	0.4634	0.2404	0.0176	0.3094	0.4494	0.6023	0.9895
$\theta_{2,2010.2}$	1.0000	73793	0.4501	0.2447	-0.0130	0.2924	0.4395	0.5974	0.9729
$\theta_{2,2010.3}$	1.0001	38772	0.4732	0.2642	-0.0091	0.2999	0.4559	0.6280	1.0530
$\theta_{2,2010.4}$	1.0000	60858	0.4473	0.2650	-0.0478	0.2750	0.4339	0.6054	1.0154
$\theta_{2,2011.1}$	1.0000	124378	0.4102	0.2634	-0.0977	0.2402	0.4035	0.5725	0.9580
$\theta_{2,2011.2}$	1.0000	105450	0.4034	0.2582	-0.0904	0.2341	0.3961	0.5639	0.9397
$\theta_{2,2011.3}$	1.0000	135419	0.3757	0.2560	-0.1257	0.2104	0.3715	0.5370	0.8970
$\theta_{2,2011.4}$	1.0000	42287	0.4163	0.2773	-0.0943	0.2331	0.4015	0.5831	1.0128
$\theta_{2,2012.1}$	1.0001	36341	0.4058	0.2718	-0.0986	0.2248	0.3923	0.5734	0.9840
$\theta_{2,2012.2}$	1.0000	55327	0.3637	0.2805	-0.1628	0.1793	0.3522	0.5354	0.9561
$\theta_{2,2012.3}$	1.0000	140767	0.2886	0.2776	-0.2550	0.1089	0.2857	0.4645	0.8533
$\theta_{2,2012.4}$	1.0000	115604	0.2152	0.2929	-0.3847	0.0325	0.2201	0.4032	0.7869
$\theta_{2,2013.1}$	1.0001	62211	0.1542	0.3064	-0.4914	-0.0314	0.1664	0.3537	0.7330
$\theta_{2,2013.2}$	1.0001	54097	0.1202	0.3202	-0.5633	-0.0725	0.1348	0.3282	0.7203
$\theta_{2,2013.3}$	1.0002	55289	0.0970	0.3195	-0.5811	-0.0959	0.1108	0.3050	0.6976
$\theta_{2,2013.4}$	1.0002	49796	0.0715	0.3108	-0.5802	-0.1198	0.0833	0.2751	0.6568
$\theta_{2,2014.1}$	1.0002	38098	0.0380	0.3075	-0.6071	-0.1532	0.0507	0.2423	0.6168
$\theta_{2,2014.2}$	1.0002	37186	0.0248	0.3121	-0.6288	-0.1693	0.0363	0.2306	0.6145

$\theta_{2,2014.3}$	1.0002	32137	0.0011	0.3250	-0.6875	-0.1978	0.0157	0.2144	0.6082
$\theta_{2,2014.4}$	1.0001	37364	0.0109	0.3316	-0.6840	-0.1905	0.0231	0.2252	0.6411
$\theta_{2,2015.1}$	1.0001	55899	0.0453	0.3195	-0.6054	-0.1541	0.0485	0.2478	0.6743
$\theta_{2,2015.2}$	1.0001	45748	0.1189	0.2956	-0.4448	-0.0751	0.1098	0.3042	0.7318
$\theta_{2,2015.3}$	1.0001	57731	0.0675	0.2812	-0.4816	-0.1164	0.0636	0.2479	0.6371
$\theta_{2,2015.4}$	1.0001	44040	-0.0204	0.2905	-0.6048	-0.2067	-0.0165	0.1691	0.5470
$\theta_{2,2016.1}$	1.0002	25878	-0.0875	0.3147	-0.7466	-0.2839	-0.0745	0.1212	0.5062
$\theta_{2,2016.2}$	1.0001	40862	-0.0677	0.3275	-0.7435	-0.2706	-0.0598	0.1421	0.5681
$\theta_{2,2016.3}$	1.0001	38333	-0.0924	0.3439	-0.8185	-0.2989	-0.0786	0.1266	0.5661
$\theta_{2,2016.4}$	1.0001	50423	-0.0842	0.3393	-0.7930	-0.2885	-0.0749	0.1296	0.5728
$\theta_{2,2017.1}$	1.0002	27934	-0.1419	0.3343	-0.8644	-0.3408	-0.1217	0.0772	0.4741
$\theta_{2,2017.2}$	1.0001	54409	-0.0873	0.3027	-0.7086	-0.2759	-0.0811	0.1070	0.5009
$\theta_{2,2017.3}$	1.0001	74176	-0.0363	0.2825	-0.5862	-0.2196	-0.0396	0.1402	0.5430
$\theta_{2,2017.4}$	1.0001	67681	-0.0249	0.2673	-0.5374	-0.1995	-0.0307	0.1415	0.5266
$\theta_{2,2018.1}$	1.0000	90295	-0.0853	0.2497	-0.5839	-0.2473	-0.0841	0.0778	0.4070
$\theta_{2,2018.2}$	1.0003	17573	-0.2057	0.2746	-0.7969	-0.3763	-0.1877	-0.0172	0.2888
$\theta_{2,2018.3}$	1.0001	30297	-0.1748	0.2801	-0.7788	-0.3438	-0.1573	0.0121	0.3364
$\theta_{2,2018.4}$	1.0001	70048	-0.1324	0.2856	-0.7368	-0.3020	-0.1209	0.0489	0.4140
$\theta_{2,2019.1}$	1.0001	91339	-0.1186	0.2950	-0.7451	-0.2888	-0.1068	0.0619	0.4540
$\theta_{2,2019.2}$	1.0000	113266	-0.1047	0.2900	-0.7218	-0.2698	-0.0950	0.0694	0.4636
$\theta_{2,2019.3}$	1.0001	73503	-0.1176	0.2802	-0.7277	-0.2718	-0.1020	0.0515	0.4137
$\theta_{2,2019.4}$	1.0002	23663	-0.1628	0.2673	-0.7897	-0.2990	-0.1322	0.0058	0.2914
$\theta_{2,2020.1}$	1.0001	40890	-0.1171	0.2118	-0.5991	-0.2302	-0.0996	0.0138	0.2631
$\theta_{2,2020.2}$	1.0000	46988	-0.0778	0.0987	-0.2752	-0.1429	-0.0763	-0.0119	0.1131
$\theta_{2,2020.3}$	1.0000	87838	-0.0542	0.1262	-0.3222	-0.1274	-0.0500	0.0231	0.1915
$\theta_{2,2020.4}$	1.0001	31429	-0.0321	0.0378	-0.1082	-0.0567	-0.0315	-0.0068	0.0402
$\theta_{2,2021.1}$	1.0008	5198	0.2557	0.2018	-0.0530	0.1047	0.2234	0.3820	0.7123
$\theta_{2,2021.2}$	1.0001	19628	0.0971	0.1121	-0.1243	0.0245	0.0977	0.1712	0.3141

Notes: Posterior estimates of the time-invariant and the time-varying parameters obtained with the shadow rate series of Wu and Xia for the period from 1999.2 to 2021.2. Priors are set according to the results in Table 1. $\theta_{1,1999.2}$ and $\theta_{2,1999.2}$ correspond to $\theta_{1,0}$ and $\theta_{2,0}$, respectively. N_{eff} stands for the effective sample size. Calculations carried out with ShinyStan.

			Krippner			Wu-X	Kia
Param.	Density	Mean	Std	[0.05, 0.95]	Mean	Std	[0.05, 0.95]
π^*	Normal	1.84	0.93	[1.69, 2.00]	1.83	0.087	[1.79, 1.98]
σ	InvGamma	0.44	0.033	[0.39, 0.50]	0.40	0.036	[0.34, 0.45]
σ_{ν_1}	InvGamma	0.083	0.034	[0.044, 0.15]	0.090	0.038	[0.046, 0.16]
σ_{ν_2}	InvGamma	0.076	0.025	[0.045, 0.12]	0.11	0.056	[0.054, 0.23]
$\theta_{1,0}$	Normal	0.57	0.30	[0.086, 1.08]	0.63	0.31	[0.14, 1.14]
$\theta_{2,0}$	Normal	0.71	0.14	[0.48, 0.94]	0.73	0.15	[0.49, 0.97]

Table B.4: Posterior distribution estimates with tight priors – 1999.2-2021.2

Notes: Posterior distribution estimates for time-invariant parameters for the period from 1999.2 to 2021.2 with tight priors for the standard deviations of the innovations in the processes governing the time-varying response to the expected inflation gap and the expected growth gap. The sample period from 1999.2 to 2001.4 was used as a training sample. The results obtained from the training sample were used as priors to estimate the model on the complete sample.

 Table B.5: Posterior distribution estimates with tight priors and without model

 training - 1999.2-2021.2

			Krippner			Wu-2	Kia
Param.	Density	Mean	Std	[0.05, 0.95]	Mean	Std	[0.05, 0.95]
$\sigma \ \pi^*$	InvGamma Normal	$\begin{array}{c} 0.46 \\ 1.84 \end{array}$	$\begin{array}{c} 0.037\\ 0.13\end{array}$	[0.40, 0.52] [1.64, 2.05]	$0.42 \\ 1.83$	$\begin{array}{c} 0.038\\ 0.12 \end{array}$	[0.36, 0.48] [1.66, 2.04]
$\theta_{1,0}$	Normal	0.35	0.41	[-0.28, 1.06]	0.44	0.43	[-0.22, 1.17]
$\theta_{2,0}$	Normal	0.59	0.25	[0.18, 1.00]	0.63	0.30	[0.16, 1.12]
$\sigma_{ u_1}$	InvGamma	0.081	0.032	[0.043, 0.14]	0.088	0.036	[0.045, 0.16]
$\sigma_{ u_2}$	InvGamma	0.076	0.026	[0.044, 0.12]	0.11	0.052	[0.052, 0.21]

Notes: Posterior distribution estimates for time-invariant parameters for the period from 1999.2 to 2021.2 with tight priors for the standard deviations of the innovations in the processes governing the time-varying response to the expected inflation gap and the expected growth gap. The estimation was carried out without conducting any training of the model.

Further results are available on demand from the author of this paper.

IMFS WORKING PAPER SERIES

Recent Issues

182 / 2023	Alexander Meyer-Gohde	Solving Linear DSGE Models with Bernoulli Iterations
181 / 2023	Brian Fabo Martina Jančoková Elisabeth Kempf Luboš Pástor	Fifty Shades of QE: Robust Evidence
180 / 2023	Alexander Dück Fabio Verona	Robust frequency-based monetary policy rules
179 / 2023	Josefine Quast Maik Wolters	The Federal Reserve's Output Gap: The Unreliability of Real-Time Reliability Tests
178 / 2023	David Finck Peter Tillmann	The Macroeconomic Effects of Global Supply Chain Disruptions
177 / 2022	Gregor Boehl	Ensemble MCMC Sampling for Robust Bayesian Inference
176 / 2022	Michael D. Bauer Carolin Pflueger Adi Sunderam	Perceptions about Monetary Policy
175 / 2022	Alexander Meyer-Gohde Ekaterina Shabalina	Estimation and Forecasting Using Mixed- Frequency DSGE Models
174 / 2022	Alexander Meyer-Gohde Johanna Saecker	Solving linear DSGE models with Newton methods
173 /2022	Helmut Siekmann	Zur Verfassungsmäßigkeit der Veranschlagung Globaler Minderausgaben
172 / 2022	Helmut Siekmann	Inflation, price stability, and monetary policy – on the legality of inflation targeting by the Eurosystem
171 / 2022	Veronika Grimm Lukas Nöh Volker Wieland	Government bond rates and interest expenditures of large euro area member states: A scenario analysis
170 / 2022	Jens Weidmann	A new age of uncertainty? Implications for monetary policy
169 / 2022	Moritz Grebe Peter Tillmann	Household Expectations and Dissent Among Policymakers

168 / 2022	Lena Dräger Michael J. Lamla Damjan Pfajfar	How to Limit the Spillover from an Inflation Surge to Inflation Expectations?
167 / 2022	Gerhard Rösl Franz Seitz	On the Stabilizing Role of Cash for Societies
166 / 2022	Eva Berger Sylwia Bialek Niklas Garnadt Veronika Grimm Lars Other Leonard Salzmann Monika Schnitzer Achim Truger Volker Wieland	A potential sudden stop of energy imports from Russia: Effects on energy security and economic output in Germany and the EU
165 / 2022	Michael D. Bauer Eric T. Swansson	A Reassessment of Monetary Policy Surprises and High-Frequency Identification
164 / 2021	Thomas Jost Karl-Heinz Tödter	Reducing sovereign debt levels in the post-Covid Eurozone with a simple deficit rule
163 / 2021	Michael D. Bauer Mikhail Chernov	Interest Rate Skewness and Biased Beliefs
162 / 2021	Magnus Reif Mewael F. Tesfaselassie Maik Wolters	Technological Growth and Hours in the Long Run: Theory and Evidence
161 / 2021	Michael Haliassos Thomas Jansson Yigitcan Karabulut	Wealth Inequality: Opportunity or Unfairness?
160 / 2021	Natascha Hinterlang Josef Hollmayr	Classification of Monetary and Fiscal Dominance Regimes using Machine Learning Techniques
159 / 2021	Volker Wieland	The decline in euro area inflation and the choice of policy strategy
158 / 2021	Matthew Agarwala Matt Burke Patrycja Klusak Moritz Kraemer Kamiar Mohaddes	Rising Temperatures, Falling Ratings: The Effect of Climate Change on Sovereign Creditworthiness
157 / 2021	Yvan Lengwiler Athanasios Orphanides	Collateral Framework:Liquidity Premia and Multiple Equilibria
156 / 2021	Gregor Boehl Cars Hommes	Rational vs. Irrational Beliefs in a Complex World
155 / 2021	Michael D. Bauer Eric T. Swanson	The Fed's Response to Economic News Explains the "Fed Information Effect"